

WILEY

Campbell Collaboration

Evidence and gap map of studies assessing the effectiveness of interventions for people with disabilities in low-and middle-income countries

Ashrita Saran¹ | Howard White¹ | Hannah Kuper²

¹Campbell Collaboration, New Delhi, India ²London School of Hygiene and Tropical Medicine (LSHTM), London, UK

Correspondence

Ashrita Saran, Evidence Synthesis Specialist, Campbell Collaboration, Global Development Network, ISID Complex, Vasant Kunj Institutional Area, New Delhi 110070, India. Email: asaran@campbellcollaboration.org

Abstract

Background: There are approximately 1 billion people in the world with some form of disability. This corresponds to approximately 15% of the world's population (World Report on Disability, 2011). The majority of people with disabilities (80%) live in lowand middle-income countries (LMICs), where disability has been shown to disproportionately affect the most disadvantaged sector of the population. Decision makers need to know what works, and what does not, to best invest limited resources aimed at improving the well-being of people with disabilities in LMICs. Systematic reviews and impact evaluations help answer this question. Improving the availability of existing evidence will help stakeholders to draw on current knowledge and to understand where new research investments can guide decision-making on appropriate use of resources. Evidence and gap maps (EGMs) contribute by showing what evidence there is, and supporting the prioritization of global evidence synthesis needs and primary data collection.

Objectives: The aim of this EGM is to identify, map and describe existing evidence of effectiveness studies and highlight gaps in evidence base for people with disabilities in LMICs. The map helps identify priority evidence gaps for systematic reviews and impact evaluations.

Methods: The EGM included impact evaluation and systematic reviews assessing the effect of interventions for people with disabilities and their families/carers. These interventions were categorized across the five components of community-based rehabilitation matrix; health, education, livelihood, social and empowerment. Included studies looked at outcomes such as, health, education, livelihoods, social inclusion and empowerment, and were published for LMICs from 2000 onwards until January 2018. The searches were conducted between February and March 2018. The EGM is presented as a matrix in which the rows are intervention categories (e.g., health) and

This is an open access article under the terms of the Creative Commons Attribution License, which permits use, distribution and reproduction in any medium, provided the original work is properly cited.

© 2020 The Authors. Campbell Systematic Reviews published by John Wiley & Sons Ltd on behalf of The Campbell Collaboration

Abbreviations: 3ie, International Initiative for Impact Evaluation; CBR, community-based rehabilitation; CEDIL, Centre for Excellence for Development Impact and Learning; DFID, Department for International Development; EGM, evidence and gap maps; GDP, gross domestic product; LMIC, low- and middle-income countries; RCT, randomised controlled trial; WHO, World Health Organisation; UNCRPD, United Nation Convention on the Rights of Persons with Disability; SDG, sustainable development goals.

Collaboration

subcategories (e.g., rehabilitation) and the column outcome domains (e.g., health) and subdomains (e.g., immunization). Each cell lists the studies for that intervention for those outcomes, with links to the available studies. Included studies were therefore mapped according to intervention and outcomes assessed and additional filters as region, population and study design were also coded. Critical appraisal of included systematic review was done using A Measurement Tool to Assess Systematic Reviews' rating scale. We also quality-rated the impact evaluation using a quality assessment tool based on various approaches to risk of bias assessment.

Results: The map includes 166 studies, of which 59 are systematic reviews and 107 impact evaluation. The included impact evaluation are predominantly quasiexperimental studies (47%). The numbers of studies published each year have increased steadily from the year 2000, with the largest number published in 2017. The studies are unevenly distributed across intervention areas. Health is the most heavily populated area of the map. A total of 118 studies of the 166 studies concern health interventions. Education is next most heavily populated with 40 studies in the education intervention/outcome sector. There are relatively few studies for livelihoods and social, and virtually none for empowerment. The most frequent outcome measures are health-related, including mental health and cognitive development (n = 93), rehabilitation (n = 32), mortality and morbidity (n = 23) and health check-up (n = 15). Very few studies measured access to assistive devices, nutrition and immunization. Over half (n = 49) the impact evaluation come from upper-middle income countries. There are also geographic gaps, most notably for low income countries (n = 9) and lower-middle income countries (n = 34). There is a fair amount of evidence from South Asia (n = 73) and Sub-Saharan Africa (n = 51). There is a significant gap with respect to study quality, especially with respect to impact evaluation. There appears to be a gap between the framing of the research, which is mostly within the medical model and not using the social model of disability.

Conclusion: Investing in interventions to improve well-being of people with disabilities will be critical to achieving the 2030 agenda for sustainable development goals. The EGM summarized here provides a starting point for researchers, decision makers and programme managers to access the available research evidence on the effectiveness of interventions for people with disabilities in LMICs in order to guide policy and programme activity, and encourage a more strategic, policy-oriented approach to setting the future research agenda.

1 | PLAIN LANGUAGE SUMMARY

1.1 | The evidence for disability interventions is unevenly distributed by sector and geography, and much of it is of poor quality

There is a considerable body of evidence related to interventions for people with disabilities and their families in low- and middle-income countries (LMICs), but it is unevenly distributed by sector and geography, and much of it is of low quality.

1.2 | What is this evidence and gap map about?

There are approximately 1 billion people in the world with some form of disability—Approximately 15% of the world's population. The majority of people with disabilities (80%) live in LMICs, disproportionately affecting the most disadvantaged sector of the population.

Decision makers need to know what works, and what does not, to best invest limited resources to improve the well-being of people with disabilities and their families in LMICs. This evidence and gap map (EGM) shows the available evidence from systematic reviews and impact evaluations.

What is the aim of this EGM?

The aim of this EGM is to show all the available evidence from systematic reviews and impact evaluations of interventions to improve the welfare of people with disabilities and their families in low- and middle-income countries (LMICs).

1.3 | What studies are included?

The EGM includes impact evaluations and systematic reviews assessing the effect of interventions for people with disabilities and their families or carers. Included studies had to report an estimate of the quantitative impact of an intervention. The studies were categorised as to whether the intervention or outcomes focused on health, education, livelihood, social inclusion or empowerment.

The map includes 166 studies: 59 systematic reviews and 107 impact evaluations. The included impact evaluations are predominantly quasiexperimental studies.

1.4 | What is the distribution of evidence?

The studies are unevenly distributed across intervention areas. Health is the most heavily populated area of the map: 118 studies of the 166 studies concern health interventions. Education is next most heavily populated (40 studies). There are relatively few studies for livelihoods and social, and virtually none for empowerment.

The most frequent outcome measures are health-related, including mental health and cognitive development (n = 93), rehabilitation (n = 32), mortality and morbidity (n = 23) and health check-up (n = 15). Very few studies measured access to assistive devices, nutrition or immunisation.

Over half (n = 49) the impact evaluations come from uppermiddle-income countries. There are also geographic gaps, most notably for low-income countries (n = 9). There is a fair amount of evidence from South Asia (n = 73) and Sub-Saharan Africa (n = 51).

There is a significant lack of high-quality studies, especially with respect to impact evaluation There also appears to be a gap in the framing of the research, which is mostly within the medical model and does not use the social model of disability. That is, the interventions mostly try to change characteristics of the person with a disability (e.g., improve social skills) rather than to address structures (e.g., readiness of schools to include people with learning disabilities).

1.5 | What do the findings of this map mean?

The EGM summarised here provides a starting point for researchers, decision makers and programme managers to access the available research evidence on the effectiveness of interventions for people with disabilities in LMICs. This EGM is important in order to guide policy and programme activity, and encourage a more strategic, policy-oriented approach to setting the future research agenda.

Campbell

Collaboration

Whilst the evidence base is relatively large, it is unevenly distributed. There is a need for more studies in rights-based approaches, livelihoods and empowerment. More studies are required from low-income settings. And study quality needs to be improved for both impact evaluations and systematic reviews.

1.6 | How up-to-date is this EGM?

The authors searched for studies published up to December 2018.

2 | EXECUTIVE SUMMARY

2.1 | Background

More than 1 billion people in the world have some form of disability. This corresponds to approxiantely 15% of the world's population (World Health Organisation, 2011). The majority of people with disabilities (80%) live in LMICs where disability has been shown to disproportionately affect the most disadvantaged. In 2004, the World Bank estimated the global gross domestic product (GDP) loss due to disability to be between \$1.71 trillion and \$2.23 trillion annually, mainly because of the exclusion of people with disabilities from employment opportunities (Metts & Mondiale, 2004).

Although disability research in LMICs is growing, several important questions have not been adequately addressed. For example, what type of evidence is needed, and what are realistic expectations, to improve outcomes and inclusion for people with disabilities? This short report summarizes preliminary findings from an EGM commissioned by Department for International Development (DFID) under the Centre for Excellence for Development Impact and Learning (CEDIL) programme, and undertaken by the Campbell Collaboration and the International Centre for Evidence and Disability.

2.2 | Objectives

The EGM presents studies of the effectiveness of interventions for people with disabilities and their families in LMICs across a range of outcome domains. Specifically, the objectives of the EGM were to:

- a. Develop a clear framework of types of interventions and outcomes related to effectiveness of interventions for people with disabilities and their families in LMICs.
- b. Map available systematic reviews and impact evaluation on the effectiveness of disability interventions in LMICs in this framework, with an overview provided in a summary report.

-WILEY

c. Provide database entries of included studies which summarize the intervention, context, study design and main findings.

2.3 | What is an EGM?

An EGM is a presentation of the available, relevant evidence for a particular sector. Relevance is defined in relation to the scope of the map. This report is for a map of studies of the effectiveness of interventions to improve the welfare of people with disabilities and their families in LMICs. The map is a table or matrix which provides a visual presentation of the evidence. In the disability map the rows are intervention categories and the columns are indicator (outcome) categories. Both interventions and indicators have subcategories.

2.4 | Search method

The search was conducted in three stages:

- 1. Populating the map based on a search of systematic reviews.
- 2. Populating the map based on search of Impact evaluation.
- 3. Populating the map based on grey literature search.

The search was carried out in February/March 2018 on: (a) academic databases, such as Medline and Web of Science; (b) international organization websites including DFID, (c) existing EGMs and (d) systematic review databases such as the Campbell Library. Only studies published since 2000 with a focus on one or more LMIC were eligible for inclusion. The search yielded over 46,000 hits, with over 35,000 hits coming from the search on OVID. One hundred sixty-six studies were included in the final map after screening and coding.

2.5 | Selection criteria

The target populations are people with disabilities and their families living in LMICs, based on the World Bank classification. According to the United Nation Convention on the Rights of Persons with Disabilities (UNCRPDs), people with disabilities include those who have long-term physical, mental, intellectual or sensory impairments, which in interaction with various barriers may hinder their full and effective participation in society on an equal basis with others. The population sub-groups of interest for this EGM include: women, vulnerable children (particularly children in care), conflict (conflict and postconflict settings), migrants and ethnic minority groups.

Studies with multiple populations are included in the map as long as they have a LMIC focus.

Reviews with a global focus are included if they did not have any search restriction excluding LMICs.

2.6 | Screening, data extraction and quality appraisal

Title and abstract screening and the evidence classification were undertaken by two independent reviewers, and any discrepancies were resolved by a third reviewer. The studies that passed on to full text were screened against the eligibility criteria by two independent reviewers, and conflicts resolved by third reviewer. After screening, all studies were coded for a wide array of information and populated into the map.

The studies were coded by the intervention category and subcategory. The intervention categories are those from the WHO community-based rehabilitation (CBR) guidelines (WHO, 2010): health, education, livelihood, social and empowerment. Advocacy and governance was added as a sixth category, given its importance to the DFID approach.

The coded information includes: bibliographic details for the study, the interventions from the framework that the study evaluates, the outcomes from the framework that the study measures and other relevant aspects such as population, region and countries. This coding was done by two independent reviewers and conflicts reconciled by a third reviewer. The quality of the included systematic reviews was assessed using A Measurement Tool to Assess Systematic Reviews (AMSTAR 2) and done independently by two reviewers. We also quality rated the impact evaluation (individual studies) based on the quality assessment tool for individual studies. This tool included six criteria (study design, sample size, attrition, definition of intervention, definition of outcome, baseline balance) that are appropriate for the assessment of quantitative impact evaluations

2.7 | Results

The map includes 166 studies, of which 59 are systematic reviews and 107 impact evaluation. The included impact evaluation are predominantly quasiexperimental studies (47%). The numbers of studies have increased fairly steadily from the year 2000 with the largest number published in the year 2017.

The studies are unevenly distributed across intervention areas. Health is the most heavily populated area of the map. A total of 118 studies of the 166 studies concern health interventions. Education is next most heavily populated with 40 studies in the education intervention/outcome sector. There are relatively few studies for livelihoods and social, and virtually none for empowerment.

The most frequent outcome measures are health-related, including mental health and cognitive development (n = 93), rehabilitation (n = 32), mortality and morbidity (n = 23) and health check-up (n = 15). Very few studies measured access to assistive devices, nutrition or immunization.

Over half (n = 49) the impact evaluation come from upper-middle income countries. There are important geographic gaps, most notably for low income countries (n = 9) and lower-middle income countries (n = 34). There is a fair amount of evidence from South Asia (n = 73) and Sub-Saharan Africa (n = 51). There is a significant gap with respect to study quality, especially with respect to impact evaluation. There appears to be a gap between the framing of the research, which appears to be mostly within the medical model (i.e., change at individual level) and not on the social model of disability (i.e., change at service or system level).

The majority of studies focus on people with physical impairments. There is a significant lack in studies focusing on people with visual or hearing impairment.

There is an important gap with respect to study quality, especially with respect to impact evaluation. Many of the included systematic reviews were assessed to have methodological limitations.

The findings from this EGM highlights a number of gaps, as mentioned above. Due to the strong concerns on the quality of reviews and impact evaluation, the evidence base needs to be strengthened on what works to improve the well-being of people with disabilities and their families in LMICs. We identify the following implications for research:

- More studies are needed to fill an important gap in measuring intervention for people with disability and incorporating considerations for equity, with increased focus on low income settings.
- 2. Ensuring that the funding and research agencies adopt best practice approach for conducting and reporting research to raise the quality of available data.

3 | BACKGROUND

3.1 | The problem, condition or issue

Disability is an umbrella term, covering impairments, activity limitations and participation restrictions. The Preamble to the UNCRPD acknowledges that disability is "an evolving concept", but also stresses that "disability results from the interaction between persons with impairments and attitudinal and environmental barriers that hinder their full and effective participation in society on an equal basis with others" (United Nations General Assembly, 2006).

Impairments can relate to vision, hearing, physical, psychosocial and cognitive or other bodily functions. An impairment becomes disabling when individuals are prevented from participating fully in society because of social, political, economic, environmental or cultural factors. This definition of disability is in line with a biopsychosocial conceptualization of disability, recognising the importance of both the impairment and contextual factors in causing difficulties in participation. This definition informs the current EGM. This approach draws on the earlier, medical model focussed more on the importance of impairments, as well as the social model which concentrates on the role of society in the exclusion of people with impairments.



More than 1 billion people in the world have some form of disability. This corresponds to approximately 15% of the world's population (World Health Organisation, 2011). The majority of people with disabilities (80%) live in LMICs and disability disproportionately affects the most disadvantaged (Banks, Kuper, & Polack, 2017). People with disabilities are more likely to experience a range of exclusions, including from employment, education, healthcare access and social participation (World Health Organisation, 2011). As a consequence, people with disabilities are more likely to live in poverty, both because disability causes poverty, but also because people who are poor are more likely to become disabled (World Health Organisation, 2011). In addition to economic impact, employment serves many nonfinancial functions. For example, at the individual level, work provides a sense of purpose and belonging in society, leading to improved self-esteem, greater autonomy and an enriched guality of life (Walsh & Tickle, 2013). More broadly, disability is linked to social exclusion and low levels of autonomy and sense of empowerment.

The link of disability and poverty is also borne at a global level, as evidenced by a large systematic review (Banks et al., 2017). In 2004, the World Bank estimated the global GDP loss due to disability to be between \$1.71 trillion and \$2.23 trillion annually (Metts & Mondiale, 2004). Turning to general disability, a World Bank study estimated that exclusion from the labour market results in a total loss of US \$891 million/year in Bangladesh and that income losses among adult caregivers add an additional loss of US\$234 million/year (World Bank, 2008).

People with disabilities are not a homogenous group, and include people with different ages, genders, impairment types and living in different settings, and this may influence the impact of disability. The systematic review showed that the link between poverty and disability is apparent for both males and females and regardless of poverty measure used (Banks et al., 2017). Poverty and disability were linked across impairment types, although a clearer link may have existed for people with mental conditions. Similarly, a study on link between poverty and disability found that people with mental illness face higher levels and intensity of poverty, partly as a result of stigma and prejudice (Trani & Loeb, 2012). There is some evidence that the relationship is strongest in countries with higher income level, that is, in upper-middle versus lower income countries. This means that as countries move out of poverty the people with disabilities are increasingly left behind. The review showed that the association of poverty and disability may be strongest in the working population age group. Similarly, another study that used internationally comparable data of 15 developing countries, found that people with disabilities aged 40 and above and people with multiple disabilities were more likely to be multidimensionally poor' (Mitra, 2013).

A key argument in attaining welfare for people with disabilities is to equalize social and economic opportunities from both humanitarian and economic perspectives. From a humanitarian perspective, interventions are implemented to secure basic human rights for people with disabilities. From an economic perspective, programmes WILEY Campbell Collaboration

are expected to increase the human capital of people with disabilities, and thus enable them to reduce their dependence on income transfers and other forms of public support. This economic expectation addresses disability as a development issue. Research is now required to determine the most cost-effective ways to overcome the above obstacles and develop policies and strategies that increase the economic contributions of people with disabilities (Metts & Mondiale, 2004).

Disability is also a human rights issue, as well as a development issue, and this is highlighted in a range of international documents, including the World Programme of Action Concerning Disabled People (WPA, 1982), the Convention on the Rights of the Child (CRC, 1989), the Standard Rules on the Equalisation of Opportunities for People with Disabilities (United Nation, 1994) and most importantly The United Nations Convention on the Rights of Persons with Disabilities (UNCRPD, 2006). The UNCRPD aims to "promote, protect and ensure the full and equal enjoyment of all human rights and fundamental freedoms by all persons with disabilities, and to promote respect for their inherent dignity". It reflects the major shift in global understanding and responses towards disability, and emphasises that people with disabilities have the right for full inclusion.

Inclusive development includes and involves everyone, especially those who are marginalized and often discriminated against (UNDP, 2010). The justification for disability inclusive development is that unless people with disabilities are brought into mainstream it is impossible to break the cycle of poverty and discrimination. Attention to disability issues is now increasingly being seen in the policies and programmes of bilateral agencies like Department of International Development (DFID, 2000) either as part of inclusive new policies or in disability-specific initiatives. Furthermore, disability is included as a specific focus within several of the sustainable development goals (SDGs). Although there is little data on the cost effectiveness of disability inclusive development, The Asian Development Bank maintains that the costs associated with including people with disabilities are far outweighed by the long-term financial benefits to individuals, families and society (ADB, 2005; Banks & Polack, 2015).

3.2 | The intervention

The "Twin-Track approach" promotes integration of disabilitysensitive measures into the design, implementation, monitoring and evaluation of all development policies and programmes, called as "mainstreaming disability", while simultaneously undertaking "targeted measures" such as disability-specific policies, programmes and initiatives to ensure the inclusion and full enjoyment of human rights by persons with disabilities (UNDP, 2010). A twin-track approach may be required to enable people with disabilities to contribute to creating opportunities, share in the benefits of development, and participate in decision-making (DFID, 2000). The twin-track approach aims to break this cycle between disability, poverty and exclusion, by both empowerment of individuals/families/organisations and by breaking down barriers in society, and is advocated for by many international donors (e.g., the World Bank, DFID, German Cooperation; the European Community [EC], the Finnish Cooperation) and non-governmental organisation (NGOs).

The World Health Organisation's (WHO) CBR guidelines is based on this approach. CBR is a multisectoral, bottom-up strategy which operates at the community level. While the UNCRPD provides the philosophy and policy of disability-inclusive development, CBR is a practical strategy for its implementation (Mendis, Gunnel, Ann, & Einar, 1989). CBR activities are designed to meet the basic needs of people with disabilities, reduce poverty, and enable access to health, education, livelihood and social opportunities—all these activities fulfil the aims of the UNCRPD.

Therefore, the CBR will serve as a guiding framework for the EGM and the five pillars of CBR: health, education, livelihood, social and empowerment will form the intervention and outcome categories.

3.3 | Why it is important to develop the EGM

Over the past decade, the academic literature on disability outcomes and effectiveness has grown substantially (Andresen, Lollar, & Meyers, 2000; lemmi et al., 2015; Ramey et al., 2016). However, several important questions have not been adequately addressed. For example, what type of evidence is needed, and what are realistic expectations, for disability inclusive interventions? A lack of rigorous and comparable data on disability and evidence on programmes that work can impede understanding and action.

Understanding the numbers of people with disabilities and their circumstances can improve efforts to remove disabling barriers and provide services to allow people with disabilities to participate on an equal basis with others. Many efforts are currently underway to fill these knowledge gaps and generate internationally comparable data on the living situation and needs of people with disabilities. This is an important step towards persuading policy and programme decision makers that disability is an issue that needs urgent attention. However, it does not help them in determining which actions are required.

Knowledge production to influence policy and programme action takes place across several sectors (health, social welfare and education), focuses on various populations (different ages, ethnicities or with different needs), and involves rather diverse methodical approaches (e.g., systematic reviews, impact evaluation of different designs etc.). A mapping of the existing knowledge base is, therefore, required to provide a comprehensive overview of existing knowledge in this area, to improve the discoverability, and thereby the use, of that evidence. Furthermore, an EGM can show implementing agencies where there is no relevant information for their programmes and enable the purposeful and targeted commissioning of future research, tailored to the most eminent needs for knowledge and guidance. The EGM can also help to identify gaps to be filled by evidence synthesis where sufficient information is available within

7 of 34

one subject area. This overview of the existing evidence is provided by the EGM presented in this study.

4 | OBJECTIVES

4.1 | Objectives

The EGM presents studies of the effectiveness of these interventions across a range of outcome domains. Specifically, the objectives of the map have been to:

- i. Develop a clear framework of types of interventions and outcomes related to effectiveness of interventions for people with disabilities and their families in LMICs.
- ii. Map available systematic reviews and impact evaluation on the effectiveness of disability interventions in LMICs in this framework, with an overview provided in a summary report.
- iii. Provide database entries of included studies which summarize the intervention, context, study design and main findings.

5 | METHODS

5.1 | EGM: Definition and purpose

EGMs provide a visual overview of the availability of evidence for a particular sector-in this case "people with disabilities and their families". The EGM consolidates what we know and do not know

about "what works" by mapping out existing and ongoing systematic reviews and impact evaluations in this field; and provides a graphical display of areas with strong, medium, weak or nonexistent evidence on the effect of interventions or initiatives.

-WILEY

Campbell

Collaboration

The EGM is presented in two dimensions: the rows list interventions and the columns list outcome domains. Each cell shows studies which contain evidence on that combination of intervention and outcomes. This EGM provides an overview of the existing systematic reviews and impact evaluations on the key outcome domains and interventions aimed to increase the welfare of people with disabilities in LMICs. EGMs show what studies are available. In accordance to the recommendations in the Campbell EGM Guidance document, maps do not summarize what the evidence says.

Impact evaluations are those intended to assess causal effects, also referred to as effectiveness studies. These studies are described in more detail in section below. Hence the map does not include the following sorts of studies: (a) prevalence studies of different impairments, (b) studies on the barriers and issues faced by people with disabilities, (c) process evaluations of interventions intended to benefit people with disabilities, and (d) ethnographic, participatory and other qualitative research or action research on people with disabilities. All of these studies are an important part of the body of research to understand and improve the lives of people with disabilities in LMICs, but they are not within the scope of this map.

The map has additional dimensions which capture study or intervention characteristics, such as study design, region, countries and population subgroup (which includes type of disability).





WILEY-Campbell Collaboration

The online version of the map (Figure 1) is interactive so that users may click on entries to see a list of studies for any cell in the map. The map is constructed using software prepared by the EPPI Centre. The cells of the table contain a bubble whose size is proportional to the number of studies reporting that outcome for that intervention. There are separate bubbles for impact evaluation and systematic reviews, with the reviews further divided by study quality. The map includes a set of filters allowing evidence to be shown just for certain sub-populations, such as specific regions or countries.

5.2 | Types of evidence

The EGM of the effectiveness of interventions for people with disabilities shows the available evidence on the success of interventions to improve the lives of people with disabilities and their families in LMICs. Included studies adopt the International Classification of Functioning, Disability and Health, medical or social model conceptualization of disability.

The EGM includes systematic reviews of effects of interventions, as well as impact evaluations that used: (a) randomised experimental design, (b) rigorous quasiexperimental design, (c) natural experiments, (d) regression discontinuity, (e) propensity score matching, (f) difference in difference, (g) instrumental variables, (h) other matching designs or (i) single-subject designs. Given the small number of studies in the map, the map also includes before versus after studies intended to address causal effects, though the absence of a comparison group means that we have low confidence in study findings from these studies.

The EGM includes both completed and on-going studies. Ongoing studies are those which are in progress or the full review is not yet published. The reference for such studies is the study protocol.

5.3 | Type of population

The target populations for this EGM are people with disabilities living in LMICs, based on the World Bank Classification (World Bank, 2017). We also included studies targeting parents/caregivers of people with disabilities. Other populations (e.g., teachers) may be targeted as a means for improving circumstances for people living with disabilities. For this map, we do not focus on the prevention of impairments.

People with disabilities include those who have long-term physical, mental, intellectual or sensory impairments, which in interaction with various barriers may hinder their full and effective participation in society on an equal basis with others (UNCRPD, 2006).

For this map we will include following type of disabilities:

 Physical: A physical impairment is the long-term loss or impairment of part of a person's body function, resulting in a limitation of physical functioning, mobility, dexterity or stamina. It will include conditions as cerebral palsy, Spina Bifida, poliomyelitis, spinal cord injuries.

- 2. Visual: Visual impairment, also known as vision impairment or vision loss, is a decreased ability to see to a degree that causes problems in daily life. Conditions may include complete or partial loss of vision, due to conditions such as macular degeneration, retinal detachment and so on.
- 3. Hearing: Hearing impairment refers to partial or total inability to hear.
- 4. Intellectual: Also known as learning disability. This condition is characterized by significantly impaired intellectual and adaptive functioning which arises before the age of 18. This involves a permanent limitation in a person's ability to learn.
- Mental: This category includes conditions such as Schizophrenia, Alzheimer's, bipolar disorders, psychosis.

If a paper includes mixture of disability types, that paper was coded for all types of disabilities as included. Similarly if the study included individual participant with multiple disabilities, again the study was coded for all those disabilities.

In recent years, the inclusion of traditionally underrepresented groups in research has received increasing attention, including racial and ethnic minorities, women, elderly individuals and children (Glickman et al., 2008). These groupings are relevant with respect to disability, as these characteristics may heighten vulnerability in the face of disability, and may also relate to a higher prevalence of disability.

Hence, the population sub-groups of interest for this EGM include: women, vulnerable children (particularly children in care), conflict (conflict and postconflict settings), migrants and ethnic minority groups.

5.4 | Types of interventions

The SDG guidelines highlight that implementing the SDGs should build upon existing international and national commitments and mechanisms, in order to generate an inclusive and global dialogue. The WHO's CBR recognizes CBR as a comprehensive and multisectoral strategy to equalize opportunities and include people with disabilities in all aspects of community life. Therefore, the CBR will serve as a guiding framework for the intervention and outcome categories as listed below, in order to realize the full inclusion and empowerment of persons with disabilities. We have added "Advocacy and Governance" as one of the components, as strong advocacy may be required to prevent and/or address abuse, neglect and exploitation that people with disabilities may experience (CBM, 2012).

The six main intervention categories are:

- 1. Health.
- 2. Education.
- 3. Livelihood.
- 4. Social.

- 5. Empowerment.
- 6. Advocacy and governance.

5.5 | Types of outcome measures

The five main outcome categories are as mentioned below and they are plotted against the WHO's CBR indicators:

- 1. Health.
- 2. Education.
- 3. Livelihood.
- 4. Social.
- 5. Empowerment.

5.6 | Types of settings

The EGM includes studies from LMICs. Studies with multiple populations are included in the map as long as they have a LMIC focus. Reviews with a global focus are included as eligible if they do not exclude countries from LMICs.

The World Bank region classification will be used as filters. There is also a filter for studies in conflict and postconflict settings.

5.7 | Search methods and sources

The EGM is based on comprehensive search for impact evaluation and systematic reviews based on the framework of interventions and outcomes as outlined above.

The Campbell Collaboration policy brief for searching studies and information retrieval, informed the search strategy as presented below (Hammerstrøm, Wade, Jørgensen, & Hammerstrøm, 2010). In addition, information retrieval specialist John Eyers was consulted during the preparation of search strings, while several search retrieval specialists provided recommendations during the peerreviewing process of the study protocol. The lead author conducted the searches once the protocol had been peer-reviewed and approved by Campbell Collaboration. The searches were conducted during the period February 19, 2018 to March 9, 2018.

At the end of the screening process, key journals were searched using key-terms up to the end of January 2018. Studies in any language and from any country were included, provided the abstract was in English.

Searches were completed, as per protocol with a number of minor additions. In some cases the search string could be copied and pasted directly from the protocol, whilst other databases required the search string to be manually populated as recommended by Higgins and Green (2011), the search strategy is reported in Appendix B. Details of additional grey literature databases are included as recommended by Campbell Collaboration information retrieval specialists.

5.8 | Electronic searches

The search was as comprehensive as possible, using (but not limited to) relevant systematic review database for first stage along with bibliographic databases (Appendix D), EGM databases, web-based search engines, websites of specialist organisations, bibliographies of relevant reviews, and targeted calls for evidence using professional networks or public calls for submission of articles. Database for EGMs was also searched to identify any map and relevant populated studies. In addition, reference lists of the included reviews were reviewed and the authors contacted for information on other relevant sources. Citation searches were also performed (see Appendix B).

Campbell Collaboration

5.9 | Searching other sources

We searched the following databases to identify unpublished reviews studies: Dissertation Abstracts, Conference Proceedings and Open Grey. We also searched a number of agency websites.

To identify ongoing studies, we searched ClinicalTrials.gov and WHO International Clinical Trials Registry Platform and CENTRAL Trials Register within the Cochrane Library for published trials.

5.10 | Stakeholder engagement

An advisory group consisting of international experts in disability contributed to the preparation of the EGM by commenting on the EGM framework and advised on dissemination strategies. Members for this advisory panel are:

5.10.1 | Professor Tom Shakespeare

He is Professor of Disability Research, London School of Hygiene & Tropical Medicine. His primary research interests are in disability studies, medical sociology, and in social and ethical aspects of genetics. He has had a long involvement with the disabled people's movement in UK and internationally. In the context of disability arts, he has also been active in arts and culture, and was a member of Arts Council England from 2003 to 2008. During his 5 years at WHO, he helped produce and launch key reports such as the World Report on Disability (World Health Organisation, 2011) and International Perspectives on Spinal Cord Injury (World Heath Organisation & International Spinal Cord Society, 2013), and was responsible for the UN statement on forced, coerced and otherwise involuntary sterilization (World Health Organisation, 2014).

5.10.2 | Dr. David Olichini

He is the Head of Prevention and Health Unit, NCDs Technical Advisor, Humanity and Inclusion Federation. He leads the elaboration of the strategic plan for the Prevention and Health's Direction including

-WILEY

-WILEY-C Campbell Collaboration

definition of Key Performance Indicators, action plan and budget allocation for each of the sectors in the PHD, namely NCDs, Mental Health, Road safety and Sexual and reproductive health and right. He is the lead for various reports and publications on various aspects of disability.

5.10.3 | Professor G. V. S. Murthy

He is the Director, IIPH Hyderabad, India. His work revolves around improving global health and fostering international partnerships to improve health status of populations. He worked at WHO, Geneva on the Childhood Blindness Program and was a UNAIDS Consultant with National Aids Control Organisation (NACO) for 2 years, where he guided and monitored the first Behavioural Surveillance Survey undertaken by NACO and facilitated the development of the first Computerized Monitoring Information System for the National AIDS Control Program. He is an international expert on public health disability and has been engaged in generating evidence on health care access and health concerns of persons with disability and in developing innovative interventions to dismantle these barriers. He has undertaken research projects in India, Bhutan, Bangladesh, Pakistan, Nigeria, Nepal and Sri Lanka. Dr. Murthy is Technical Advisor on Disability to CBM South Asia, Technical Advisor (Research) for Mission for Vision. He is a member of the Queen Elizabeth Diamond Jubilee Trust's Scientific Advisory Committee, National Task Force on DR and ROP, Optometry Council of India, IAPB DR Technical Advisory Committee.

6 | DIMENSIONS

6.1 | Scope

The WHO's CBR programme recognizes CBR as a comprehensive and multisectoral strategy to equalize opportunities and include people with disabilities in all aspects of community life. CBR activities are designed to meet the basic needs of people with disabilities, reduce poverty, and enable access to health, education, livelihood and social opportunities—all these activities fulfil the aims of the UNCRPD. Therefore, the CBR will serve as a guiding framework and the five pillars of CBR: health, education, livelihood, social and empowerment will form the intervention and outcome categories. Clinical/pharmacological interventions to prevent or treat the primary impairment/health condition are beyond the scope of the map and hence such studies are excluded. We will be including studies that focus specifically on people with disabilities, as well as studies referring to interventions for families of people with disabilities.

6.2 | Conceptual framework

The matrix (Figure 1) illustrates the different sectors, which can make up a CBR strategy for the welfare of people with disabilities. It consists of five key components, each divided into five key elements. The elements are subdivided into content headings. The matrix should not be seen as sequential, and all components will not be needed by every person with disabilities Figure 2.

6.3 | Description of intervention/problem categories

The included interventions cover all main strategies to reduce disability related outcome as described in the CBR. The six main intervention categories are:

- 1. Health.
- 2. Education.
- 3. Livelihood.



FIGURE 2 Community-based rehabilitation matrix

- 4. Social.
- 5. Empowerment.
- 6. Advocacy and Governance (added-not part of CBR matrix).

6.4 | Description of population/geographic location/outcome categories

The five main outcome categories are as mentioned below and they are plotted against the WHO's CBR indicators:

- 1. Health.
- 2. Education.
- 3. Livelihood.
- 4. Social.
- 5. Empowerment.

6.5 | Description of population/geographic location

The EGM has two primary dimensions: interventions (rows) and outcomes (columns).

Additional dimensions are:

- (1) Population subgroups of interest include: age group (underfive, children, adolescent and elderly), women, vulnerable children (particularly children in care), conflict (conflict and postconflict settings), migrants and ethnic minority groups.
- (2) Study designs: The EGM includes systematic reviews of effects of interventions and impact evaluations that used one of: (a) randomised experimental design, (b) quasiexperimental design (controlled before-after and uncontrolled before-after), (c) natural experiments, (d) regression discontinuity, (e) propensity score matching, (f) difference in difference, (g) instrumental variables and (i) single-subject designs.
- (3) World Bank region: South Asia, Sub-Saharan Africa, East Asia and Pacific, Latin America and Caribbean, Middle East and North Africa, Europe and Central Asia.
- (4) Type of impairment/disability: Physical impairment, visual impairment, mental impairment, hearing impairment, intellectual/learning impairment.
- (5) Conflict-affected regions.
- (6) Economies: Low-income economies, lower-middle-income economies, upper-middle-income economies, high-income economies.

7 | DATA COLLECTION AND ANALYSIS

7.1 | Screening and study selection

An information specialist validated the detailed search strategy developed by the team covering a combination of academic databases, organisational websites and grey literature. Detailed

search strategy is provided in the online protocol (link). All search results were imported in to EPPI reviewer for screening and coding.

The screening of studies in relation to inclusion/exclusion was undertaken in two stages. The first stage involved title and abstract; the second involved full text documents.

Three independent researchers were involved at each stage. The screening was carried out based on predefined eligibility criteria (Appendix A) by two independent reviewers and the third screener resolved the conflicts. Prior to data extraction and coding, the three independent reviewers met to discuss and pilot the extraction and coding procedures on a sample of abstracts.

Stage 1: Screen on Title and abstract

The screening was carried out based on predefined eligibility criteria (Appendix A) by two independent reviewers and the third screener resolved the conflicts. The conflict was resolved by third reviewer through group discussion with team. Title and abstracts which passed the first stage were retrieved in full text for a more comprehensive review.

Stage 2: Screen on full text

Full text documents were retrieved for all documents that passed stage one. Two reviewers independently evaluated all studies. Studies had to meet all of the inclusion/exclusion criteria set out previously in order to advance to full review.

7.2 | Data extraction and management

Each included study was coded independently by two coders using the coding tool covering study characteristics, population, intervention, outcomes, region, countries and type of disability. The coding tool is added in the Appendix C.

7.3 | Tools for assessing risk of bias/study quality of included reviews

Each study in the map has a rating for the quality of evidence.

For systematic reviews, we scored each study using the 16-item checklist called AMSTAR 2 ("Assessing the Methodological Quality of Systematic Reviews"). The 16 items cover: (1) PICOS in inclusion criteria, (2) ex ante protocol, (3) rationale for included study designs, (4) comprehensive literature search, (5) duplicate screening, (6) duplicate data extraction, (7) list of excluded studies with justification, (8) adequate description of included studies, (9) adequate risk of bias assessment, (10) report sources of funding, (11) appropriate use of meta-analysis, (12) risk of bias assessment for meta-analysis, (13) allowance for risk of bias in discussing findings, (14) analysis of heterogeneity, (15) analysis of publication bias and (16) report conflicts of interest.

Items 2, 4, 7, 9, 11, 13 and 15 are termed "critical". Study quality is rated high if there is no more than one noncritical weakness, and medium if there is no critical weakness but more than one non critical weakness. Studies with one or more critical weaknesses are rated low quality. WILEY Compbell

Impact evaluation: The quality assessment for the impact evaluations is based on existing approaches to risk of bias assessment. Many of the items in this assessment, such as study design and baseline balance, relate to possible sources of bias. Other items relate to clarity of reporting, especially of the intervention and outcomes. The assessment used the following criteria (see Tables 1-3):

- Study design (potential confounders taken into account): Impact evaluations need either a well-designed control group, preferably based on random assignment, or an estimation technique which controls for confounding and the associated possibility of selection bias.
- Adequate sample size: Small samples generally mean that a study in underpowered, that is, there is a high risk of not finding an effect even if the intervention works.
- 3. Attrition (or loss to follow-up) can be a major source of bias in studies, especially if these is differential attrition between the treatment and comparison group so that the two may no longer be balanced in preintervention characteristics. The US Institute of Education Sciences What Works Clearing House (Deke, Emily Sama-Miller, & Alan Hershey, 2015) has developed standards for acceptable levels of attrition, in aggregate and the differential, which are applied here.
- 4. Clear definition of disability: For a study to be useful the study population must be clear, which means that the type and degree of disability should be clearly defined, preferably with reference to a widely-used international standard (e.g., Washington Group questions).
- 5. Clear definition of outcome measures is needed in order to aid interpretation and reliability of findings and comparability with other studies. Studies should clearly state the outcomes being used with a definition and the basis on which they are measured, preferably with reference to a widely-used international standard.
- 6. Baseline balance shows that the treatment and comparison groups are the same at baseline. Lack of balance can bias the results.

Overall study quality is the lowest rating awarded any one of the above six criteria.

8 | RESULTS

8.1 | Description of studies

8.1.1 | Results of the search

The search yielded over 46,000 hits, with over 35,000 hits coming from the search on OVID. Given the large number of hits, text mining on abstracts was used to narrow down the search results. EPPI reviewer uses machine-learning algorithm (text-mining) to prioritize the order in which references are presented for screening. The ranking of references continuously improves as screening progresses and more manual decisions are available from which the algorithm can learn. In EPPI Reviewer, citations are ranked in their order of relevance after choosing "starts priority screening". This process fastens the screening process and left us with 9,842 hits, of which 237 were duplicates, leaving 9,606 studies for title and abstract screening. Of these, 547 studies were identified for full text screening and 100 were eligible for inclusion.

Phase 2 involved back screening included studies from systematic reviews and grey literature search. In Phase 2, an additional 35 studies were identified through back referencing. Grey literature search was performed and an additional 31 studies were included from grey literature. See the PRISMA flow chart in Figure 3.

As a result of this process, a total of 166 studies were included for coding. Of these, 59 are systematic reviews (see references for a list of included studies), and 107 impact evaluations. We then screened the included studies in the 59 systematic reviews to assess their eligibility for inclusion in the map.

8.2 | Synthesis of included studies

Studies in the map, especially systematic reviews, may be coded under more than one intervention category or subcategory. This means that there are many more entries in the map then there are studies. The number of studies contained in the map is stated clearly at the top of the map.

8.3 | Risk of bias in included reviews

Figure 4 shows the results of the critical appraisal of the 107 included impact evaluations. Three quarters of the impact evaluations (78%) are rated as low confidence in study findings.

The high figure of low confidence is largely driven by concerns related to sample size and attrition. Seventy-four percent of the impact evaluations had sample size <30. Attrition is not reported in half of the included studies.

8.4 | Synthesis of included studies

8.4.1 | Publication of studies over time

Figure 5 shows number of studies evaluating the effects of disability interventions published each year between 2000 and 2018. Since 2000 there has been a gradual increase in the number of studies from 1 to 21 new studies published in the year 2017. The search was conducted until end of January 2018 and hence it did not capture the studies published after January 2018. The majority of studies consistently focussed on health over the years.

| ARAN ET AL. | | Campbell -WILEY 13 of 3 |
|------------------------------------|--|--|
| ABLE 1 Lists the interv | vention subcategories under each of the | ese headings |
| CBR pillar (intervention category) | Component (intervention subcategory) | Examples |
| Health | Promotion Prevention | Parent/family training and education, inclusive health promotion campaigns, health care provider training Introduction of specific intervention measures through better nutritional practices; improvement of health services, early detection and diagnosis; prenatal and postnatal care |
| | Medical care Rehabilitation | Periodic health screening, access to routine healthcare Access to specialist care, such as physiotherapy, speech and language therapy, occupational therapy; cognitive stimulation, rehabilitation and training, activity therapy centres, supportive therapy, stress-management interventions/psychosocial support, trauma informed therapy |
| | Assistive devices | Provision of appliances (orthoses, prostheses, hearing aids, etc.), devices such as day calendars with symbol pictures for people with cognitive impairment, communication boards and speech synthesizers for people with speech impairment |
| Education | Early childhood Primary secondary and higher Nonformal Lifelong Learning | Early intervention (e.g., play therapy), preschool/kindergarten provision Inclusive childhood education Provision of learning material and special equipment (Braille, audio cassettes, sign language, etc.), recruitment and training of specialized teachers, resource rooms, bypass intervention Faith-based schools, home-based learning, play groups Adult literacy programs, continuing education, life and survival skills |
| Livelihood | Skills development Self-employment Financial services Waged employment Social protection | Training opportunities for jobs, home-based trainings, vocational training, training in mainstream institutions and community based trainings Income generation program Access to credit Quota legislation in jobs and participation in labour intensive public works programs Social insurance schemes, social assistance intervention |
| Social | Relationship, marriage and family Personal assistance Culture and arts | Support in role as parents, protection from violence, building awareness in community of rights of people with disabilities to a family life Provision, training and support of informal and formal personal assistance Promoting use of art for social change like positive portrayal, silent theatres, complementary therapy in the form of art, music. Inclusive art education, diversity trainings, encouraging inclusion in mainstream cultural programmes, work with spiritual and religious leaders and groups. |
| | Recreation, leisure and sports Access to justice | Provision of adapted sports equipment, organization of inclusive sports events, linking people with disabilities to mainstream recreation and sporting clubs/associations, positive media coverage of disability recreation, using recreation Legal awareness, Identification of available resources like local leaders, legal centres, legal aid. Promoting legal rights and empowerment, inheritance right, community or legal aid centre |
| Empowerment | Communication | Improve access to information and communication resources; reduce |

communication barriers and improve representation for people with disabilities; Strengthen communication skills of CBR personnel

Building trust and credibility within community, raise awareness in the community, motivate the community to participate, bringing stakeholder

Reservation of position in public and political institution, promoting political awareness and access to political process, disability awareness

Creating joint resources like training material, community directories,

advocating rights of persons with disability, partnership with existing self-

Establishment/reinforcement of a special education service in the Ministry of Education Establishment/Reinforcement of medical rehabilitation centres Legislative reforms: elimination of all forms of discrimination raising awareness on human rights through media appropriate budgetary allocation

together, capacity building, celebrating achievements

within political system

help groups

Abbreviation: CBR, community-based rehabilitation.

Advocacy and Governance

Social mobilisation

Political participation

organisations (DPOs)

Self-help groups and disabled people's

TABLE 2 List of outcome categories and subcategories

| Outcome | WHO's community-based rehabilitation (CBR) indicators |
|--|--|
| Health component | |
| Mental health and cognitive development | People with disabilities equally access mental health services and engage in activities needed to achieve the highest attainable standard of mental health services |
| Access to health services | People with disabilities equally access health services and engage in activities needed to achieve the highest attainable standard of health Percentage of people with disabilities and their families that have access to medical care |
| | People with disabilities feel they are respected and treated with dignity when receiving health services |
| Immunization | Percentage of people with disabilities who receive full immunization as recommended for their country by WHO |
| Health check-up | People with disabilities know how to achieve good levels of health and participate in activities contributing to their health Percentage of children with disability who receive the recommended health check-ups |
| Rehabilitation services | People with disabilities have access to, and use rehabilitation services |
| Access to assistive devices | People with disabilities have access to use, and know how to maintain appropriate assistive products in their daily life |
| Nutrition | People with disabilities have access to nutritional support to maintain a healthy diet |
| Morbidity and mortality | People with disabilities access and benefit from quality medical services appropriate to their life stage needs and priorities |
| Education | |
| Enrolment to primary, secondary and tertiary education | Policies and resources are conducive to quality education for people with disabilities and ensure smooth transitions through different stages of learning People with disabilities participate in and complete quality primary education in an enabling and supportive environment People with disabilities have resources and support to enrol and complete quality secondary and higher education in an enabling and supportive environment |
| Attendance | People with disabilities experience post school options on an equal basis with their peers People with disabilities have resources and support to enrol and complete quality |
| Education in mainstream education facilities/ | Percentage of people with disabilities who acquire education in mainstream education |
| inclusive education | facilities |
| Social and life skill development | People with disabilities make use of youth or adult centered learning opportunities to improve their life skills and living conditions |
| Learning and achievement | People with disabilities experience equal opportunities to participate in learning opportunities that meet their needs and respect their rights |
| Access to educational services | People with disabilities participate in a variety of nonformal learning opportunities based on their needs and desires People with disabilities actively participate in early childhood developmental activities and |
| | play, either in a formal or informal environment |
| Livelihood | |
| Employment in formal and informal sector | People with disabilities have paid and decent work in the formal and informal sector on equal bases with others |
| | People with disabilities earn income through their own chosen economic activities People with disabilities acquire marketable skills on an equal basis with others through a range of inclusive training opportunities |
| Access to job market | People with disabilities have access to job markets on equal basis as others |
| Control over own money | People with disabilities have control over the money they earn |
| Access to financial services such as grants and loans | People with disabilities have access to grants, loans and other financial services on an equal basis with others |
| | People with disabilities participate in local saving and credit schemes |
| Poverty and out-of-pocket payment | Percentage of people with disabilities who are covered by social protection programs |
| Access to social protection programs | People with disabilities access formal and informal social protection measures they need |
| Participation in development of inclusive policies | Inclusive policies, practices and appropriate resources, defined with people with disabilities enable equal participation of women and men with disability in livelihood (training, finance, work opportunities and social protection) |
| Social | |
| Stigma and discrimination | Communities have increased awareness about disability, with a reduction in stigma and discrimination towards people with disabilities People with disabilities feel safe in their family and community |
| Participation in mainstream recreational, leisure and sports activity | People with disabilities participate in inclusive or specific recreation, leisure and sports activities |

TABLE 2 (Continued)

| Outcome | WHO's community-based rehabilitation (CBR) indicators |
|--|--|
| Legal rights Access to justice Participation in cultural and religious activity Interpersonal interaction and relationships | All People with disabilities are recognized as equal citizens with legal capacity People with disabilities access and use formal and informal mechanisms of justice People with disabilities participate in artistic, cultural or religious events in and outside their home as they choose People with disabilities experience support of the community and their families to socialize |
| Social identity and responsibilities | and form age-appropriate and respectful relationships Percentage of people with disabilities who feel respected in their decisions regarding personal relationships People with disabilities feel valued as community members and have a variety of social identities, roles and responsibilities |
| Empowerment | |
| Informed choices | People with disabilities make informed choices and decisions |
| Positions in public institutions and judiciary | People with disabilities participate in political processes on an equal basis with others |
| Voting rights | People with disabilities participate in political processes on an equal basis with others |
| Representation at community level | People with disabilities actively engage in and benefit from self-help groups in the local communities, if they choose (inclusive or specific) |
| | Self-help groups come together to form federations to harness collective energy and influence positive change |
| | People with disabilities living in different situations (rural or urban areas, poor or rich, refugees) feel they are adequately represented by DPO |
| Advocacy | People with disabilities effectively use communication skills and resources (including supportive decision making) to facilitate interactions and influence change |
| | People with disabilities play a catalyzing role in mobilizing key community stakeholders to create an enabling environment |

Abbreviations: DPO, disabled people's organisation; WHO, World Health Organisation.

| TABLE 3 | Quality | assessment | of | impact | evaluation |
|---------|---------|------------|----|--------|------------|
|---------|---------|------------|----|--------|------------|

| lte | m | Point in time (where applicable) | Rating |
|-----|---|----------------------------------|---|
| 1 | Study design (potential confounders taken into account) | End of intervention | High confidence: Randomised controlled trial (RCT), regression discontinuity design, interrupted time series, instrumental variable Medium confidence: Difference in differences with matching, propensity score matching Low confidence: Other matching |
| 2 | Adequate sample size | | High confidence: Sample size ≥100 or cluster ≥60 Medium confidence: Sample size <100 or cluster <60 Low confidence no power calculation or sample size <30 or cluster <30 |
| 3 | Losses to follow up are presented and acceptable ^a | End of intervention | High: attrition within Institute of Education Science (IES) bounds ^a Medium: attrition close to IES bounds Low: attrition not reported or attrition outside IES bounds N/A for ex post studies |
| 4 | Intervention if clearly defined | | High confidence: intervention clearly and fully described Medium confidence: brief description of intervention Low confidence: intervention named but not described, or not named |
| 5 | Outcome measures are clearly defined and reliable | | High confidence: outcome measure clearly and fully described, preferably with reference to validation Medium confidence: brief description of outcome Low confidence: outcome named but not described |
| 6 | Baseline balance (not applicable for before versus after) | | High confidence: RCT or baseline balance report and satisfactory (imbalance on 5 or less than 5 percent) Medium confidence: Imbalance between 5-10 percent Low confidence: Baseline balance not reported, or reported and lack of balance on 10 or more than 10 percent |
| | Overall confidence in study findings | End of intervention | Lowest rating across items 1a, 4a, 6 and 7 |

^ahttps://homvee.acf.hhs.gov/HomVEE-Attrition-White_Paper-7-2015.pdf.



FIGURE 3 PRISMA for disability evidence and gap map

Figure 6 shows the quality trends of both impact evaluations and systematic reviews over the years. The proportional number of low quality impact evaluations has increased over the years as compared to medium/high quality impact evaluations. A high proportion of systematic reviews identified had methodological limitations and were of low quality.

8.5 | By intervention categories

Systematic reviews are concentrated in the health sector: 45 (80%) of reviews report effects of health interventions (Figure 7). Randomised controlled trials (RCTs) account for close to half of the impact evaluations (44 RCTs out of 107) being particularly prominent in health and education (30 and 13 studies, respectively), where some studies cover both sectors.



FIGURE 4 Number of impact evaluation by study quality

As mentioned above a single study may appear in more than one category. For example, Velema, Ebenso, and Fuzikawa (2008) review of rehabilitation programmes states that the interventions covered include "home visits by trained community workers who taught disabled persons skills to carry out activities of daily living, encouraged disabled children to go to school, helped find employment or an income generating activity, often involving vocational training and/or microcredit. Many programmes had a component of influencing community attitudes towards disabled persons". This study is coded under each of health, education, livelihoods, empowerment and advocacy and governance intervention type.

8.6 | By type of impairment

Nearly two-thirds of the studies (60%) of the studies relate to interventions for people with mental or intellectual impairments, 27% to physical impairment, with a small number identified as relating to hearing and visual and hearing impairments (see Figure 8; recall that some studies are coded under more than one category).

8.7 | By outcome domain

Since the most common intervention category is health, it is unsurprising that the health-related outcomes are reported in the largest number of studies (114 studies); see Figure 9. This is followed by education (46), social (46) and livelihoods (24 studies). Only 3 included studies report empowerment-related outcomes. Systematic reviews are concentrated in the health sector: 46 (78%) of reviews report effects of health interventions (Figure 8). RCTs being particularly prominent in health sector (37) and considerably less than 10 RCTs in other sectors.

Within health, mental health and cognitive development account for the largest number of studies (93 studies) followed by rehabilitation (32) (Figure 10).

8.8 | By region

Impact evaluations are unevenly distributed across World Bank region and countries (Table 4). Over half the impact evaluation come from four LMICs. These are concentrated in four countries: India (23), China (11), Iran and Turkey (9) studies each. South Asia is relatively well covered with studies from India (23), Bangladesh (5) and Pakistan (4) and as is East Asia on account of China (Figures 11-13).

We included in the map all reviews in which studies from LMICs were eligible in searches. However, only 17 of the 59 included reviews actually include eligible studies. Of the other 44 reviews, 20 studies had only included studies from high-income countries, 16 had LMIC studies which were not eligible for reasons of date or study design, 5 had no included studies and 2 are ongoing with results not yet reported.

Thirty-eight impact evaluations concerned fragile and conflicted affected states (Table 5).



FIGURE 5 Trends in publication of studies by intervention





8.9 | By quality assessment

The systematic reviews were assessed using the AMSTAR tool described elsewhere in the document. Of the 59 reviews, 22 were assessed as low quality, 16 medium and 18 high quality, with the remaining three studies ongoing and, therefore, not yet scorable.



FIGURE 7 Number of studies by study design and intervention categories



FIGURE 8 Number of studies relating to type of impairment

8.10 | By population

Figure summarises the number of Impact evaluations by population. Fifty-three impact evaluations focussed on children and sixteen on adults. Nine impact evaluations were identified that focused on interventions for parent/caregivers and teachers. There were limited impact evaluation on vulnerable groups and ethnic minorities.



FIGURE 9 Number of studies by study design and outcome domain



FIGURE 10 Number of studies by health outcomes

| TABLE 4 | Countries | in impact | evaluation | by | income | group |
|---------|-----------|-----------|------------|----|--------|-------|
|---------|-----------|-----------|------------|----|--------|-------|

| Low income | | Lower-middle income | | Upper-middle income | |
|------------|---|---------------------|----|---------------------|----|
| N = 9 | | N = 50 | | N = 45 | |
| Rwanda | 1 | Armenia | 1 | Turkey | 9 |
| Uganda | 3 | Indonesia | 2 | Iran | 9 |
| Ethiopia | 3 | Kenya | 4 | China | 11 |
| Eritrea | 1 | Bangladesh | 5 | Lebanon | 1 |
| Togo | 1 | Egypt, Arab Rep | 3 | Brazil | 3 |
| | | India | 23 | South Africa | 5 |
| | | Pakistan | 4 | Thailand | 3 |
| | | Nigeria | 3 | Russia | 1 |
| | | Vietnam | 2 | Peru | 1 |
| | | Zambia | 2 | Malaysia | 2 |
| | | Ukraine | 1 | | |

8.11 | Analysis by CBR pillars

8.11.1 | Health

The health quadrant of the EGM—which map the studies of the effects of health interventions on health outcomes—is the most heavily populated section of the map (Table 6). The total in Table 6 exceeds 166 (the total number of studies) as many studies are coded under more than one intervention and outcome.



FIGURE 11 Countries with largest number of impact evaluations



FIGURE 12 Completed Impact evaluations by population subgroup



FIGURE 13 Evidence architecture. Source: White (2019)

Mental health is prominent in this quadrant: 93 studies report outcomes for mental health and cognitive development. Indeed, mental health dominates the map with the three largest bubbles being health intervention studies—medical care, rehabilitation and promotion—reporting a mental health outcome measure. Other heavily populated cells in the health domain are the rehabilitation outcomes from rehabilitation interventions, with an additional five reporting rehabilitation outcomes from health promotion

TABLE 5 Impact evaluation from fragile and conflicted-affected states (*N* = 38)

| High fragility | 7 |
|--------------------|----|
| Pakistan | 3 |
| Nigeria | 3 |
| Eritrea | 1 |
| Moderate fragility | 13 |
| Lebanon | 1 |
| Iran | 9 |
| Egypt | 3 |
| Low fragility | 11 |
| Ethiopia | 3 |
| Bangladesh | 5 |
| Kenya | 3 |
| Neighbours | 7 |
| Zambia | 1 |
| Uganda | 3 |
| Thailand | 3 |
| | |

TABLE 6 Aggregate map: Number of studies by intervention category and outcome (impact evaluation/reviews)

| | | Intervention | Interventions | | | | |
|----------|------------------|--------------|---------------|------------|--------|-------------|-------|
| | | Health | Education | Livelihood | Social | Empowerment | Total |
| Outcomes | Health | 58/44 | 9/10 | 6/5 | 11/17 | 1/0 | 161 |
| | Education | 10/9 | 21/11 | 2/3 | 6/8 | 1/1 | 67 |
| | Livelihood | 6/6 | 0/2 | 9/10 | 3/2 | 1/1 | 39 |
| | Social | 17/14 | 7/7 | 3/1 | 14/12 | 1/1 | 77 |
| | Empower- ment | 1/2 | 0/0 | 0/0 | 1/1 | 1/2 | 7 |
| | Total | 166 | 66 | 38 | 71 | 10 | 351 |

interventions. Morbidity and mortality outcomes are also quite well represented with 25 studies, mainly from medical care and rehabilitation interventions.

Health interventions are generally the most heavily represented across nonhealth outcomes. Most notably there are 11 studies of rehabilitation interventions under the "social outcome" indicator interpersonal interaction and relationships.

8.11.2 | Education

Whilst 40 studies are classified in the intervention category and 46 studies have education-related outcomes, closer analysis of these figures is needed to see the distribution of studies.

The studies classified under education very largely do not refer to participation of children with disabilities in formal education. The most commonly reported education outcome is "social and life skills development" (36 studies) with effects reported from health interventions (rehabilitation and promotion), as well as early child development, and nonformal education. On the intervention side, there are equal studies for nonformal education (18 studies) and primary/secondary (19 studies) and early child development (14).

8.11.3 | Livelihoods

Out of 22 included studies for livelihood interventions, 16 studies focused on skill development. On the outcome side 24 studies included livelihood outcomes. These two groups do not necessarily overlap; there are studies reporting the impact of livelihood interventions on health and livelihood outcomes, and there are studies analysing the impact of health interventions on livelihood outcomes. However, the studies included in the reviews are mostly of low quality and hence the relevant conclusions as made based on them can be undermined.20172011

8.11.4 | Social and empowerment

Out of 35 studies included for social interventions 18 studies focus on relationship, marriage and family, followed by personal sparse in the areas assessing the impact on stigma and discrimination (7), safety (4), participation in mainstream recreational, leisure and sports activity (2). Similarly very few studies are included for interventions as access to justice (4) and sports and recreation and leisure (6).

assistance interventions (16). On the outcome side 39 studies

included interpersonal interaction and relationship. Evidence is

Campbell Collaboration

9 | DISCUSSION AND GAPS IN EVIDENCE

Findings of this map are primarily helpful for researchers, policy makers and development practitioners that require evidence to inform policy and practice. National government and international partners can use it to identify existing evidence related to intervention of interest. Researcher and funders can identify areas suitable for evidence synthesis and move away from areas which may be saturated and also help explicitly identify gaps in knowledge.

EGMs are important building blocks in the evidence architecture and help in the following three ways:

- 1. Guide users to high quality reviews.
- 2. Guide users to where there are no high quality reviews.
- Guide users to evidence gaps to be filled by new reviews and impact evaluations.

9.1 | Areas of high quality reviews and impact evaluations

9.1.1 | Systematic reviews

Out of the 18 high quality review identified for this map, 13 (73%) are in health sector. A large proportion of these focuses on rehabilitation and health promotion.

Some high quality reviews were also identified in education (5) and social (5) pillars and may have some policy implications. Within education, high quality reviews were identified on early childhood Education and Non-formal education. While only one high quality review was identified in the primary and secondary education.

A significant number of reviews were found to have methodological limitation, particularly in empowerment and livelihood sectors.

9.1.2 | Impact evaluation

WILEY

20 of 34

Out of 13 high quality impact evaluations, 11 were identified in the health sector (85%). Even within health, there was unevenly distributed between subcategories; rehabilitation (6), medical care (4), promotion (3) and prevention (1). No high quality impact evaluations were identified on assistive devices.

Campbell

High quality impact evaluations were scarce in other sectors and only one high quality impaction was identified in education, livelihood and social sector. No high quality impact evaluations are available on empowerment.

9.1.3 | Areas of major gaps in the evidence

We now summarise the important evidence gaps based on the analysis of included impact evaluations and systematic reviews in the map.

Many areas of the map are sparsely populated or unevenly distributed with evidence. The most evident gaps relate to empowerment and advocacy interventions and empowerment outcomes. For livelihood intervention, most of the areas are scarcely populated. There is only one study on self-employment and limited studies on waged employment, financial services and social protection. For social interventions, sport, recreation and leisure and access to justice are scarcely populated.

Even where there are pockets of evidence, such as health and education, more studies are still needed. For instance in health, there are only few studies assessing effectiveness of assistive devices. Similarly in education, there are few studies on lifelong learning.

Following the same pattern, certain outcomes have received more attention than others. Health outcomes are certainly the most studies outcomes but gaps exist within this as well. Only three studies assessed access to assistive devices. Significant gaps exist in empowerment and livelihood outcomes. Within livelihood, outcome such as control over own money, poverty and out-of-pocket payments, access to social protection programs, participation in development of inclusive policies were least studied. Though social outcomes were fairly concentrated, limited studies were identified assessing stigma and discrimination, safety, participation in mainstream leisure and sports activity, legal rights, access to justice, participation in cultural and religious activities.

Most of the studies come from upper middle income countries, and even within this almost all the evidence comes from four countries, China, Turkey, Iran and South Africa. Similarly, in lowermiddle income countries most of the studies were undertaken in three countries:India (40), Pakistan (13) and Bangladesh (12).

There are also very few studies from low-income countries, reflecting the relative neglect of many parts of Sub-Saharan Africa such as Rwanda, South Sudan, Somalia, Congo, Burundi and so on. Studies were scarce for people with disabilities in many of these conflict affected regions. There were also gaps by impairment type, with limited studies identified on people with visual and hearing impairment.

There is mostly low or medium confidence in study findings, and so another gap is the absence of high quality studies in the field. Reviews are of higher quality overall, though less than one-third qualified as high quality and the studies they draw on tend to be of low quality.

We can draw on two Rapid Evidence Assessments undertaken from the EGM, and thereby go beyond the bounds of what reading the map can tell us (Kuper, Saran, & White, 2018; White, Saran, & Kuper, 2018). It is apparent from the reviews that the focus of studies is on fixing individuals, that is, the medical approach, for instance, focussing on improving social or learning skills for people with disabilities. Fewer studies focus on improving infrastructure or institutions, and therefore address social barriers to inclusion. Development agencies, including DFID, are stressing the biopsychosocial approach in their work, so the absence of evidence on what works to promote disability inclusion is a very striking gap. Future systematic reviews from the EGM will be able to provide more guidance for action for policy and programme decision-makers.

9.1.4 | Limitations of the EGM

- The EGM provide a rich source of information on existing systematic reviews and impact evaluations relating to interventions to improve the lives of people with disabilities and their families in LMICs.
- The EGM followed comprehensive search using predefined eligibility criteria, yet inevitably there are limitations to our approach.
- Eligible studies were restricted to those published after 2000 up until the start of 2018, and published in English. Also searching the "grey" literature is challenging, and consequently some eligible studies may have been missed.
- Sometimes it was difficult for the reviewer team to categorize interventions, mainly between empowerment and livelihood, as there can be overlaps. The categorization for such interventions was done based on expert consultations and the information as available to mitigate this issue as far as possible.

10 | AUTHORS' CONCLUSIONS

The mapping exercise is a first step to identifying priority areas for systematic reviews and impact evaluations. We identify initial steps that can help advance research to promote the welfare and inclusion of people with disabilities. We strongly believe that the online interactive visualization, list of references, and summary of studies will facilitate access and use of research.

10.1 | Implications for research, practice and/or policy

- The available high and medium quality systematic reviews in health sector may suggest some implications for policy. However, few of the studies are recent, and so they may need to be updated.
- Efforts are also needed to reach a consensus to identify priority areas for research with weak evidence synthesis by key funders and researchers in the field.
- More studies should be carried out given the relative lack of impact evaluation in many areas such as empowerment and livelihood. Impact evaluations will be more useful if they focus on more diverse set of outcomes and thereby fill multiple evidence gaps.
- More studies are needed to fill important gaps in equity and measuring interventions for vulnerable populations. This includes areas of gender, ethnic minorities and low-income and conflict affected settings.
- The geographical base of evidence needs to be expanded as well. Most of the studies come from upper-middle income countries and there is limited evidence from low income countries. Evidence need to be expanded in these countries.
- More studies are needed to generate evidence on all types of impairment, including visual and hearing impairments.
- Future research should also follow the best practice and improve reporting of intervention implementation in order to improve the quality of studies.
- Consideration needs to be given to improve quality of systematic reviews in terms of reporting and inclusion criteria or scope by adherence to standard guidelines as PRISMA.
- The future research agenda should explicitly consider the possibility for analyzing rights-based approaches. A variety of evaluation designs might be appropriate: such as cluster randomization for community-based approaches, and encouragement designs for national initiatives to promote inclusiveness.

11 | INFORMATION ABOUT THIS EGM

11.1 | EGM authors

11.1.1 | Lead EGM author

The lead author is the person who develops and co-ordinates the EGM team, discusses and assigns roles for individual members of the team, liaises with the editorial base and takes responsibility for the on-going updates of the EGM.

11.2 | Contributions of authors

Campbell Collaboration

Content expertise:

Dr Hannah Kuper, Director of the International Centre for Evidence in Disability, a research group at LSHTM that works to expand the research and teaching activities of LSHTM in the field of global disability. Her main research interest is disability in low and middle income countries, with a particular focus on assessment of the prevalence of disability and impairments, including in children, and development of new methods in undertaking these surveys (e.g., use of mobile technologies), investigation of the health and rehabilitation needs of people with disabilities, and how these can be met in low resources settings and research on the relationship between poverty and disability, and the potential role of social protection in breaking this cycle. She has an undergraduate degree from Oxford University in Human Sciences and a doctorate from Harvard University in epidemiology. She has worked at LSHTM since 2002.

11.2.1 | Systematic review method expertise

All authors are experienced systematic reviewers, which means they are proficient in carrying out the various processes in an EGM, such as eligibility screening, quality assessment and coding.

EGM methods expertise:

All team members have previous experience in systematic review methodology, including search, data collection, statistical analysis, theory-based synthesis, which mean they are proficient in carrying out the various processes in an EGM, such as search, eligibility screening, quality assessment and coding.

Information retrieval expertise:

All authors have previous experience in developing search strategies.

ACKNOWLEDGEMENTS

We would like to acknowledge UK Department of International Development (DFID) under its support for the Centre for Excellence for Development Impact and Learning (CEDIL and PENDA grant from DFID for the support.

CONFLICT OF INTEREST

No conflict of interests.

PLANS FOR UPDATING THE EGM

Provided the availability of funding the EGM will be updated annually by lead author or co-authors.

-WILEY

DIFFERENCES BETWEEN PROTOCOL AND MAP

Collaboration

LEY-

None.

SOURCES OF SUPPORT

This EGM is supported by the UK Department of International Development (DFID) under its support for the Centre for Excellence for Development Impact and Learning (CEDIL). Dr Hannah Kuper is supported by the PENDA grant from DFID.

REFERENCES

References to included studies

- Abd-Elaziz, S. A., Khedr, E. M., Ahmed, H. A. E., & Ibrahim, H. D. F. (2015). Effect of cognitive rehabilitation on improving cognitive function and activities of daily living among elderly patients with stroke at Assiut University Hospital. *Journal of Education and Practice*, 6(24), 44–56.
- Acil, A. A., Dogan, S., & Dogan, O. (2008). The effects of physical exercises to mental state and quality of life in patients with schizophrenia. *Journal of Psychiatric and Mental Health Nursing*, 15(10), 808–815.
- Ahmadi, A., Arastoo, A. A., & Nikbakht, M. (2010). The effects of a treadmill training programme on balance, speed and endurance walking, fatigue and quality of life in people with multiple sclerosis. *International SportMed Journal*, 11(4), 389–397.
- Ahmed, N., Gandhi, S., & Baruah, A. (2015). Effectiveness of specific intervention on treatment adherence of persons with mental illness: A pilot study. *Indian Journal of Psychiatry*, 57(4), 403–406.
- Akbari, E., Asemi, Z., Daneshvar Kakhaki, R., Bahmani, F., Kouchaki, E., Tamtaji, O. R., ... Salami, M. (2016). Effect of probiotic supplementation on cognitive function and metabolic status in Alzheimer's disease: A randomized, double-blind and controlled trial. *Frontiers in Aging Neuroscience*, *8*, 256.
- Akbari, B., Teymori, Z., Abolghasemi, S., & Khorshidiyan, H. (2013). Stress coping strategies in hearing-impaired students. *Auditory and Vestibular Research*, 22(1), 41–49.
- Ali, B. S., Rahbar, M. H., Naeem, S., Gul, A., Mubeen, S., & Iqbal, A. (2003). The effectiveness of counseling on anxiety and depression by minimally trained counselors: A randomized controlled trial. *American Journal of Psychotherapy*, 57(3), 324–336.
- Almerie, M. Q., Okba Al Marhi, M., Jawoosh, M., Alsabbagh, M., Matar, H. E., Maayan, N., & Bergman, H. (2015). Social skills programmes for schizophrenia. *Cochrane Database of Systematic Reviews*, CD009006.
- Angba, T. O. (2016). Aerobic exercise responses and blood pressure measurement of individuals with intellectual disability in Ibadan. *Journal of Education and Practice*, 7(22), 110–114.
- Asher, L., Patel, V., & De Silva, M. J. (2017). Community-based psychosocial interventions for people with schizophrenia in low and middle-income countries: systematic review and meta-analysis. BMC Psychiatry, 17(1), 355.
- Au, M. K., Chan, W. M., Lee, L., Chen, T. M., Chau, R. M., & Pang, M. Y. (2014). Core stability exercise is as effective as task-oriented motor training in improving motor proficiency in children with developmental coordination disorder: a randomized controlled pilot study. *Clinical Rehabilitation*, 28(10), 992–1003.
- Awada, G. M., & Gutierrez-Colon, M. (2017). Effect of inclusion versus segregation on reading comprehension of EFL learners with dyslexia: Case of Lebanon. *English Language Teaching*, 10(9), 49–60.

- Bangpan, M., Dickson, K., Felix, L., & Chiumento, A. (2017). The impact of mental health and psychosocial support interventions on people affected by humanitarian emergencies: A systematic review humanitarian evidence programme. Humanitarian Evidence Programme. Oxford: Oxfam GB.
- Banks, L. M., Kuper, H., & Polack, S. (2017). Disability and social protection programmes in low- and middle-income countries: a systematic review. Oxford Development Studies, 45(3), 223–239.
- Bass, J., Neugebauer, R., Clougherty, K. F., Verdeli, H., Wickramaratne, P., Ndogoni, L., ... Bolton, P. (2006). Group interpersonal psychotherapy for depression in rural Uganda: 6-month outcomes: randomised controlled trial. *The British Journal of Psychiatry*, 188(6), 567–573.
- Bass, J., Poudyal, B., Tol, W., Murray, L., Nadison, M., & Bolton, P. (2012). A controlled trial of problem-solving counseling for war-affected adults in Aceh, Indonesia. Social Psychiatry and Psychiatric Epidemiology, 47(2), 279–291.
- Berhane, G, Devereux, S, Hoddinott, J, Hoel, J, Nega, T, Fredu, ... Schwab, B. (2015). Evaluation of the Social Cash Transfers Pilot Programme Tigray Region, Ethiopia: Endline Report. International Food Policy Research Institute, Institute of Development Studies, University of Sussex, and University of Mekelle, Tigray.
- Besler, F., & Kurt, O. (2016). Effectiveness of video modeling provided by mothers in teaching play skills to children with autism. *Educational Sciences: Theory and Practice*, 16(1), 209–230.
- Bhatia, T., Mazumdar, S., Mishra, N. N., Gur, R. E., Gur, R. C., Nimgaonkar, V. L., & Deshpande, S. N. (2014). Protocol to evaluate the impact of yoga supplementation on cognitive function in schizophrenia: A randomised controlled trial. Acta Neuropsychiatrica, 26(5), 280–290.
- Biggeri, M., Deepak, S., Mauro, V., Trani, J. F., Kumar, J., Ramasamy, P., & Giriyappa, R. (2008). Impact of CBR: Community-based rehabilitation programme in Mandya district (Karnataka, India). *Disability and Rehabilitation*, 36(18).
- Bonfioli, E., Berti, L., Goss, C., Muraro, F., & Burti, L. (2012). Health promotion lifestyle interventions for weight management in psychosis: A systematic review and meta-analysis of randomised controlled trials. *BMC Psychiatry*, 12(1), 78.
- Bordbar, M. R. F., Soltanifar, A., & Talaei, A. (2009). Short-term familyfocused psycho-educational program for bipolar mood disorder in Mashhad. Iranian. *Journal of Medical Sciences*, 34(2), 104–109.
- Botha, U. A., Koen, L., Galal, U., Jordaan, E., & Niehaus, D. J. (2014). The rise of assertive community interventions in South Africa: A randomized control trial assessing the impact of a modified assertive intervention on readmission rates; a three year follow-up. *BMC Psychiatry*, 14(1), 56.
- Botha, U. A., Koen, L., Joska, J. A., Hering, L. M., & Oosthuizen, P. P. (2010). Assessing the efficacy of a modified assertive community-based treatment programme in a developing country. *BMC Psychiatry*, 10(1), 73–73.
- Briand, C., Durand, M. J., St-Arnaud, L., & Corbiere, M. (2008). How well do return-to-work interventions for musculoskeletal conditions address the multicausality of work disability? *Journal of Occupational Rehabilitation*, 18(2), 207–217.
- Cai, J., Zhu, Y., Zhang, W., Wang, Y., & Zhang, C. (2015). Comprehensive family therapy: An effective approach for cognitive rehabilitation in schizophrenia. *Neuropsychiatric Disease and Treatment*, 11, 1247–1253.
- Carew, M. T., Deluca, M., Groce, N., & Kett, M. (2018). The impact of an inclusive education intervention on teacher preparedness to educate children with disabilities within the Lakes Region of Kenya. *International Journal of Inclusive Education*, 1–16.
- Carter, JA. 2003. Evaluation of a new instrument to assess the impact of a community-based intervention for children with communication disabilities in Kenya: Institute of Child Health, (ICH) University of London and Kenya Medical Research Institute.(KEMRI).
- Cattik, M., & Odluyurt, S. (2017). The effectiveness of the smart boardbased small-group graduated guidance instruction on digital gaming and observational learning skills of children with autism spectrum

disorder. Turkish Online Journal of Educational Technology, 16(4), 84-102.

- Chakraborty, N., Sinha, B. N. P., Nizamie, S. H., & Sinha, V. K. (2014). Evaluation of continuing nursing education (CNE) program on care and management of children with intellectual disability (India).
- Chatterjee, S., Naik, S., John, S., Dabholkar, H., Balaji, M., Koschorke, M., ... Thornicroft, G. (2014). Effectiveness of a community-based intervention for people with schizophrenia and their caregivers in India (COPSI): A randomised controlled trial. *The Lancet*, 383(9926), 1385–1394.
- Chatterjee, S., Patel, V., Chatterjee, A., & Weiss, H. A. (2003). Evaluation of a community-based rehabilitation model for chronic schizophrenia in rural India. *The British Journal of Psychiatry*, 182(1), 57–62.
- Chatterjee, S., Pillai, A., Jain, S., Cohen, A., & Patel, V. (2009). Outcomes of people with psychotic disorders in a community-based rehabilitation programme in rural India. *The British Journal of Psychiatry*, 195(5), 433–439.
- Chinchai, P., Bunyamark, T., & Sirisatayawong, P. (2010). Effects of caregiver education in stroke rehabilitation on the quality of life of stroke survivors. World Federation of Occupational Therapists Bulletin, 61(1), 56–63.
- Christopher, M., Holly, M., & Tom, S. (2014). A systematic review of the effectiveness of interventions to prevent and respond to violence against persons with disabilities. *Journal of Interpersonal Violence*, 29(17), 3207–3226.
- Cogo-Moreira, H., Andriolo, R. B., Yazigi, L., Ploubidis, G. B., Brandão de Ávila, C. R., & Mari, J J. (2012). Music education for improving reading skills in children and adolescents with dyslexia. *Cochrane Database of Systematic Reviews*, (8), CD009133.
- Coren, E., Hutchfield, J., Thomae, M., & Gustafsson, C. (2010). Parenttraining interventions to support intellectually-disabled parents. *Cochrane Database System Reviews*, (6), CD007987.
- Cui, Y., Yang, W., & Weng, Y. (2004). Effectiveness of Social Skill Training in Patients with Chronic Schizophrenia. *Chinese Mental Health Journal*, 18, 799–801.
- de Jong, K., Knipscheer, J. W., Ford, N., & Kleber, R. J. (2014). The efficacy of psychosocial interventions for adults in contexts of ongoing manmade violence—A systematic review. *Health*, 6(06), 504–516.
- Daniels, R., Metzelthin, S., van Rossum, E., de Witte, L., & van den Heuvel, W. (2010). Interventions to prevent disability in frail communitydwelling older persons: an overview. *European Journal of Ageing*, 7(1), 37–55.
- De Vibe, M. F., Bjørndal, A., Fattah, S., Dyrdal, G. M., Halland, E., & Tanner-Smith, E. E. (2017). Mindfulness-based stress reduction (MBSR) for improving health, quality of life and social functioning in adults: a systematic review and meta-analysis. *Campbell Systematic Reviews*, 13(1), 1–264.
- Devries, K., Kuper, H., Knight, L., Allen, E., Kyegombe, N., Banks, L. M., ... Naker, D. (2017). Reducing physical violence toward primary school students with disabilities. *Journal of Adolescent Health*, 62(3), 303–310.
- Dias, A., Dewey, M. E., D'Souza, J., Dhume, R., Motghare, D. D., Shaji, K. S., ... Patel, V. (2008). The effectiveness of a home care program for supporting caregivers of persons with dementia in developing countries: A randomised controlled trial from Goa, India. *PLoS One*, 3(6), e2333.
- Dickman, B, Roux, A, Manson, S, Douglas, G, & Shabalala, N. 2006. How could she possibly manage in court?'An intervention programme assisting com-plainants with intellectual disability in sexual assault cases in the Western Cape: HSRC PRESS. 2006.
- Einfeld, S. L., Stancliffe, R. J., Gray, K. M., Sofronoff, K., Rice, L., Emerson, E., & Yasamy, M. T. (2011). Interventions provided by parents for children with intellectual disabilities in low and middle income countries. *Journal of Applied Research in Intellectual Disabilities*, 25(2), 135–42.
- Eniola, M. S., & Adebiyi, K. (2007). Emotional intelligence and goal setting —An investigation into interventions to increase motivation to work

among visually impaired students in Nigeria. British Journal of Visual Impairment, 25(3), 249–253.

Campbell Collaboration-WILEY

- Faizah, R. (2018). The effectiveness of patience training to improve the quality of life of people with disabilities in rural Indonesia. *Researchers World*, 9(1), 179.
- Faulkner, G., Cohn, T., & Remington, G. (2007). Interventions to reduce weight gain in schizophrenia. *The Cochrane Library*, 33, 654-656.
- Fleming, P, Furlong, M, McGilloway, S, Keogh, F, Hernon, M, & Stainton, T. (2016). Personal budgeting interventions to improve health and social care outcomes for people with a disability. Campbell systematic reviews.
- Gabriele, J. M., Dubbert, P. M., & Reeves, R. R. (2009). Efficacy of behavioural interventions in managing atypical antipsychotic weight gain. *Obesity Reviews*, 10(4), 442–455.
- Gavrilova, S. I., Ferri, C. P., Mikhaylova, N., Sokolova, O., Banerjee, S., & Prince, M. (2009). Helping carers to care—The 10/66 dementia research group's randomized control trial of a caregiver intervention in Russia. International Journal of Geriatric Psychiatry, 24(4), 347–354.
- Gensby, U., Lund, T., Kowalski, K., Saidj, M., Jørgensen, A. M. K., Filges, T., ... Labriola, M. (2012). Workplace disability management programs promoting return to work: A systematic review. Campbell Systematic Reviews.
- Haile, G. (2011). 2011 Eritrea: Evaluation Report on the Donkey for School Project in Eritrea July 2009 - December 2010. UNICEF.
- Grider, J., & Wydick, B. (2016). Wheels of fortune: The economic impacts of wheelchair provision in Ethiopia. *Journal of Development Effectiveness*, 8(1), 44–66.
- Guerra, M., Ferri, C. P., Fonseca, M., Banerjee, S., & Prince, M. (2011). Helping carers to care: The 10/66 Dementia Research Group's randomized control trial of a caregiver intervention in Peru. *Revista Brasileira de Psiquiatria*, 33(1), 47–54.
- Gülengül, F. 2014. Effect of visual aids on learning language in children with hearing impairment. Hacettepe Üniversitesi.
- Zhanf, G. F., Tsui, C. M., Lu, A. J. B., Tsang, H. W. H., & Li, D. (2017). Integrated supported employment for people With Schizophrenia in Mainland China: A randomized controlled trial. *American Journal of Occupational Therapy*, 71, 1–8.
- Habibzadeh, H., Gofranipoor, F., & Ahmadi, F. (2007). A study on the effect of self care plan on activity daily living status in patient with cerebro vascular accident. *Journal of Medical Sciences*, 7(1), 126–130.
- Hansen, C. H., Mahmud, I., & Bhuiyan, A. J. (2007). Vocational reintegration of people with spinal cord lesion in Bangladesh: an observational study based on a vocational training project at CRP. *Asia Pacific Disability Rehabilitation Journal*, 18(1), 63–75.
- Hartling, L., Milne, A., Tjosvold, L., Wrightson, D., Gallivan, J., & Newton, A. S. (2014). A systematic review of interventions to support siblings of children with chronic illness or disability. *Journal of Paediatrics and Child Health*, 50(10), E26–E38.
- Hastings, R. P., Robertson, J., & Yasamy, M. (2012). Interventions for children with pervasive developmental disorders in low and middle income countries. *Journal of Applied Research in Intellectual Disabilities*, 25, 119–134.
- Hegde, S., Rao, S. L., Raguram, A., & Gangadhar, B. N. (2012). Addition of home-based cognitive retraining to treatment as usual in first episode schizophrenia patients: A randomized controlled study. *Indian Journal* of *Psychiatry*, 54(1), 15–22.
- Helander, E. (1989). Training in the community for people with disabilities: training package for a family member of a person who has difficulty moving (10). World Health Organisation.
- Hung, W. W. Y., & Pang, M. Y. (2010). Effects of group-based versus individual-based exercise training on motor performance in children with developmental coordination disorder: A randomized controlled pilot study. *Journal of Rehabilitation Medicine*, 42(2), 122–128.
- Iemmi, V., Blanchet, K., Gibson, L. J., Kumar, K. S., Rath, S., Hartley, S., & Kuper, H. (2015). Community-based rehabilitation for people with

-WILEY-C Campbell Collaboration

physical and mental disabilities in low-and middle-income countries: A systematic review and meta-analysis. *Journal of Development Effectiveness*, 8(3), 368–387.

- Jangid, N., Swadia, H., & Sharma, D. (2016). Effectiveness of mnemonic instructions on thinking strategies of learning disabled children. *Journal of Disability Management and Rehabilitation*, 2(1).
- Jaywant, M. S. S., & Pai, A. V. (2011). Home based program for elderly disabled, evaluation & intervention at community based centre. *Indian Journal of Occupational Therapy*, 43(3).
- John, T., Jasmine, V., Rajinikanth, & Hema (2008). Study to determine the effectiveness of group therapy for social phobia in children with learning disability. *Indian Journal of Occupational Therapy*, 40(1), 13–16.
- Kalgotra, R., & Warwal, J. S. (2017). Effect of music intervention on the behaviour disorders of children with intellectual disability using strategies from applied behaviour analysis. *Disability, CBR and Inclusive Development, 28*(1), 161–177.
- Karaaslan, O., & Mahoney, G. (2013). Effectiveness of responsive teaching with children with Down syndrome. *Intellectual and Developmental Disabilities*, 51(6), 458–469.
- Karande, S., Mehta, V., & Kulkarni, M. (2007). Impact of an education program on parental knowledge of specific learning disability. *Indian Journal of Medical Sciences*, 61(7), 398–406.
- Katongo, E. M., & Ndhlovu, D. (2015). The role of music in speech intelligibility of learners with Post Lingual Hearing Impairment in selected units in Lusaka District. Universal Journal of Educational Research, 3(5), 328–335.
- Kaur, T., Kohli, T., & Devi, B. (2008). Impact of various instructional strategies for enhancing mathematical skills of learning disabled children. Journal of Indian Association for Child and Adolescent Mental Health, 4(1), 16–19.
- Khan, F., Amatya, B., Gosney, J., Rathore, F. A., & Burkle, F. M. (2015). Medical rehabilitation in natural disasters: a review. Archives of Physical Medicine and Rehabilitation, 96(9), 1709–1727.
- Kumar, R., & Chaturvedi, S. (2014). Effectiveness of computer assisted instructional package as remedial teaching for learning disabled children. *Learning Community: An International Journal of Educational* and Social Development, 5(2/3), 163.
- Kuper, H., Saran, A., White, H., Kumar, S. T., Tolin, L., Muthuvel, T., & Wapling, L. (2018). Rapid Evidence Assessment (REA) of What Works to Improve Educational Outcomes for People with Disabilities in Low-and Middle-Income Countries. Campbell Collaboration: New Delhi, and and International Centre for Evidence on Disability: London.
- Lakhan, R. (2014). Behavioral management in children with intellectual disabilities in a resource-poor setting in Barwani, India. *Indian Journal* of Psychiatry, 56, 39–45.
- Lal, R., & Bali, M. (2007). Effect of visual strategies on development of communication skills in children with autism. Asia Pacific Disability Rehbiliation Journal, 18(2), 120–130.
- Lata, D., & Stepanyan, A. (2013). Enabling childhoods: A unique opportunity for early childhood in Armenia. *International Journal of Inclusive Education*, 17(4), 406–419.
- Lawang, W., Horey, D., Blackford, J., Sunsern, R., & Riewpaiboon, W. (2013). Support interventions for caregivers of physically disabled adults: A systematic review. *Nursing and Health Sciences*, 15(4), 534–545.
- Li, Y., Reinhardt, J. D., Gosney, J. E., Zhang, X., Xi, Hu, Chen, S., ... Li, J. (2012). Evaluation of functional outcomes of physical rehabilitation and medical complications in spinal cord injury victims of the Sichuan earthquake. *Journal of Rehabilitation Medicine*, 44(7), 534–540.
- Liebel, D. V., Friedman, B., Watson, N. M., & Powers, B. A. (2008). Review of nurse home visiting interventions for community-dwelling older persons with existing disability. *Medical Care Research and Review*, 66(2), 119–146.

- Lietz, P., Kos, J., Dix, K., Trevitt, J., Uljarevic, M., & O'Grady, E. (2018). Interventions for anxiety in school-aged children with autism spectrum disorder (ASD).
- Loh, C., Meyer, J. M., & Leckband, S. G. (2006). A comprehensive review of behavioral interventions for weight management in schizophrenia. *Annals of Clinical Psychiatry*, 18(1), 23–31.
- Lund, C., Oosthuizen, P., Flisher, A. J., Emsley, R., Stein, D. J., Botha, U., ... Joska, J. (2010). Pathways to inpatient mental health care among people with schizophrenia spectrum disorders in South Africa. *Psychiatric Services*, 61(3), 235–240.
- Lund, C., Waruguru, M., Kingori, J., Kippen-Wood, S., Breuer, E., Mannarathd, S., & Raja, S. (2013). Outcomes of the mental health and development model in rural Kenya: A 2-year prospective cohort intervention study. *International Health*, 5(1), 43–50.
- Madhavan, K., Vijayakumaran, P., Ramachandran, L., Manickam, C., Rajmohan, R., Mathew, J., & Krishnamurthy, P. (2007). Sustainable leprosy related disability care within integrated general health services: findings from Salem District, India. *Leprosy Review*, 78(4), 353–361.
- Mahakud, G. C. H., & Jena, S. P. K. (2012). Effects of cognitive-behavioural intervention for the children with reading disabilities. *Learning Community-An International Journal of Educational and Social Development*, 3(1), 89–100.
- Makanya, M. W., Runo, M., & Wawire, V. (2014). Effectiveness of transitional and follow-up programmes to community integration of young adults with intellectual disabilities (YAWID) in Kiambu County, Kenya. Journal of the American Academy of Special Education Professionals, 87–106.
- Martin, D. S., Craft, A., & Sheng, Z. N. (2001). The impact of cognitive strategy instruction on deaf learners: An international comparative study. American Annals of the Deaf, 146(4), 366–78.
- Maulik, P. K., & Darmstadt, G. L. (2007). Childhood disability in low-and middle-income countries: overview of screening, prevention, services, legislation, and epidemiology. *Pediatrics*, 120(Suppl 1), S1–S55.
- McConachie, H., Huq, S., Munir, S., Ferdous, S., Zaman, S., & Khan, N. Z. (2000). A randomized controlled trial of alternative modes of service provision to young children with cerebral palsy in Bangladesh. *The Journal of Pediatrics*, 137(6), 769–776.
- Methapatara, W., & Srisurapanont, M. (2011). Pedometer walking plus motivational interviewing program for Thai schizophrenic patients with obesity or overweight: A 12-week, randomized, controlled trial. *Psychiatry and Clinical Neurosciences*, 65(4), 374–380.
- Miyahara, M., Hillier, S. L., Pridham, L., & Nakagawa, S. (2017). Taskoriented interventions for children with developmental co-ordination disorder. *Cochrane Database of Systematic Reviews*, 7, CD010914.
- Mohammed, A. A., & Kanpolat, Y. E. (2010). Effectiveness of computerassisted instruction on enhancing the classification skill in secondgraders at risk for learning disabilities. *Electronic Journal of Research in Educational Psychology*, 8(3), 1115–1130.
- Momin, A. K. M. (2004). Impact of services for people with spinal cord lesion on economic participation. Asia Pacific Disability Rehabilitation Journal, 15(2), 53–67.
- Mourad, A. E. (2009). The effectiveness of a program based on selfregulated strategy development on the writing skills of writingdisabled secondary school students. *Electronic Journal of Research in Educational Psychology*, 7(1), 5–24.
- Mulat, M., Lehtomäki, E., & Savolainen, H. (2018). Academic achievement and self-concept of deaf and hard-of-hearing and hearing students transitioning from the first to second cycle of primary school in Ethiopia. *International Journal of Inclusive Education*, 1–15.
- Neuner, F., Schauer, M., Klaschik, C., Karunakara, U., & Elbert, T. (2004). A comparison of narrative exposure therapy, supportive counseling, and psychoeducation for treating posttraumatic stress disorder in an african refugee settlement. *Journal of Consulting and Clinical Psychology*, 72(4), 579–587.

- Nilchian, F., Shakibaei, F., & Jarah, Z. (2017). Evaluation of visual pedagogy in dental check-ups and preventive practices among 6-12-year-old children with autism. *Journal of Autism & Developmental Disorders*, 47(3), 858–864.
- Nunnerley, J., Dunn, J., McPherson, K., Hooper, G., & Woodfield, T. (2015). Participation and quality of life outcomes among individuals with earthquake-related physical disability: A systematic review. *Journal of Rehabilitation Medicine*, 47(5), 385–393.
- Nuri, M. R. P., Hoque, M. T., Akand, M. M. K., & Waldron, S. M. (2012). Impact assessment of a vocational training programme for persons with disabilities in Bangladesh. *Disability, CBR and Inclusive Development, 23*(3), 76–89.
- Oono, I. P., Honey, E. J., & McConachie, H. (2013). Parent-mediated early intervention for young children with autismspectrumdisorders (ASD). *Evidence-Based Child Health: A Cochrane Review Journal*, 8(6), 2380–2479.
- Othman, A., Ramlee, F., Ghazalan, S. A. M., Wee, K. W., & Shahidi, R. M. (2015). Effects of a brief cognitive-behavioural and social skills group intervention for children with psychological problem: A pilot study. *International Medical Journal*, 22(4), 241–245.
- ÖZcan, C. T., Oflaz, F., TÜRkbay, T., Freeman, C., & Sharon, M. (2013). The effectiveness of an interpersonal cognitive problem-solving strategy on behavior and emotional problems in children with attention deficit hyperactivity. Archives of Neuropsychiatry/Noropsikiatri Arsivi, 50(3), 244–251.
- Pajareya, K., & Nopmaneejumruslers, K. (2011). A pilot randomized controlled trial of DIR/Floortime™ parent training intervention for pre-school children with autistic spectrum disorders. Autism, 15(5), 563–577.
- Pan, R., Zhou, M., Cai, H., Guo, Y., Zhan, L., Li, M., ... Chen (2018). A randomized controlled trial of a modified wheelchair arm-support to reduce shoulder pain in stroke patients. *Clinical Rehabilitation*, 32(1), 37–47.
- Paradise, M., Cooper, C., & Livingston, G. (2009). Systematic review of the effect of education on survival in Alzheimer's disease. *International Psychogeriatrics*, 21(1), 25–32.
- Parente, M., Tofani, M., De Santis, R., Esposito, G., Santilli, V., & Galeoto, G. (2017). The role of the occupational therapist in disaster areas: Systematic review. Occupational Therapy International, 2017, 2017–8.
- Patel, V., Araya, R., Chatterjee, S., Chisholm, D., Cohen, A., De, S., ... Mark (2007). Treatment and prevention of mental disorders in low-income and middle-income countries. *The Lancet*, 370, 991–1005.
- Patel, V., Weiss, H. A., Chowdhary, N., Naik, S., Pednekar, S., Chatterjee, S., ... King, M. (2010). Effectiveness of an intervention led by lay health counsellors for depressive and anxiety disorders in primary care in Goa, India (MANAS): A cluster randomised controlled trial. *The Lancet*, 376(9758), 2086–2095.
- Pearsall, R., Smith, D. J., Pelosi, A., & Geddes, J. (2014). Exercise therapy in adults with serious mental illness: A systematic review and metaanalysis. BMC Psychiatry, 14(1), 117.
- Pedrosa, V., Pontes, A., Pontes, P., Behlau, M., & Peccin, S. M. (2016). The effectiveness of the Comprehensive Voice Rehabilitation Program compared with the vocal function exercises method in behavioral dysphonia: A randomized clinical trial. *Journal of Voice*, 30(3), 377–e11.
- Pennington, L., Parker, N. K., Kelly, H., & Miller, N. (2016). Speech therapy for children with dysarthria acquired before three years of age (Review). Cochrane Database of Systematic Reviews, 7, 2016.
- Pereira-Guizzo, C. D. S., Del Prette, A., & Del Prette, Z. A. P. (2012). Evaluation of a professional social skills program for unemployed people with physical disability. *Psicologia: Reflexão e Crítica*, 25(2), 265–274.
- Pop, C. A., Simut, R. E., Pintea, S., Saldien, J., Rusu, A. S., Vanderfaeillie, J., ... Vanderborght, B. (2013). Social robots vs. computer display: Does the way social stories are delivered make a difference for their effectiveness on ASD children? *Journal of Educational Computing Research*, 49(3), 381–401.

- Qureshi, A, Saadat, S, & Qureshi, H. 2017. Effectiveness of guided tooth brushing program for children with visual impairments-a randomized controlled trial. Biomedical Research.
- Raghavan, R., Newell, R., Waseem, F., & Small, N. (2009). A randomized controlled trial of a specialist liaison worker model for young people with intellectual disabilities with challenging behaviour and mental health needs. *Journal of Applied Research in Intellectual Disabilities*, 22(3), 256–263.
- Ran, M. S., Xiang, M. Z., Chan, C. L. W., Leff, J., Simpson, P., Huang, M. S., ... Li, S. G. (2003). Effectiveness of psychoeducational intervention for rural Chinese families experiencing schizophrenia. *Social Psychiatry* and Psychiatric Epidemiology, 38(2), 69–75.
- Reichow, B, Barton, E E, Boyd, B A, & Hume, K. 2012. Early intensive behavioral intervention (EIBI) for young children with autism spectrum disorders (ASD). Cochrane Database of Systematic Reviews (10).
- Reichow, B., Steiner, A. M., & Volkmar, F. (2012). Social skills groups for people aged 6 to 21 with autism spectrum disorders (ASD). *Campbell Systematic Reviews*, 8, 1–76.
- Reichrath, E., de, W., Luc, P., & Winkens, I. (2010). Interventions in general education for students with disabilities: a systematic review. *International Journal of Inclusive Education*, 14(6), 563–580.
- Reinhardt, J. D., Li, J., Gosney, J., Rathore, F. A., Haig, A. J., Marx, M., & Delisa, J. A. (2011). Disability and health-related rehabilitation in international disaster relief. *Global Health Action*, 4(1), 7191.
- Rezaiyan, A., Mohammadi, E., & Fallah, P. A. (2007). Effect of computer game intervention on the attention capacity of mentally retarded children. *International Journal of Nursing Practice*, 13(5), 284–288.
- Robertson, J., Emerson, E., Hatton, C., & Yasamy, M. (2012). Efficacy of community-based rehabilitation for children with or at significant risk of intellectual disabilities in low- and middle-income countries: A review. Journal of Applied Research in Intellectual Disabilities, 25, 143–154.
- Rohwerder, B. 2017. Impact of childhood screening for disability/ impairment on education and learning.
- Sani-Bozkurt, S., & Ozen, A. (2015). Effectiveness and efficiency of peer and adult models used in video modeling in teaching pretend play skills to children with autism spectrum disorder. *Education and Training in Autism and Developmental Disabilities*, 50(1), 71–83.
- Sasmal, S. (2018). Effect of an information booklet, on the knowledge and attitude regarding learning disabilities of children among the trainee teachers of selected government teachers' training institutes of Kolkata. *Nursing Journal of India*, 109(2), 77–80.
- Sharifi, V., Tehranidoost, M., Yunesian, M., Amini, H., Mohammadi, M., & Roudsari, M. J. (2012). Effectiveness of a low-intensity home-based aftercare for patients with severe mental disorders: A 12-month randomized controlled study. *Community Mental Health Journal*, 48(6), 766–770.
- Shin, J. Y., Nhan, N. V., Lee, S. B., Crittenden, K. S., Flory, M., & Hong, H. T. D. (2009). The effects of a home-based intervention for young children with intellectual disabilities in Vietnam. *Journal of Intellectual Disability Research*, 53(4), 339–352.
- Sijbrandij, M., Farooq, S., Bryant, R. A., Dawson, K., Hamdani, S. U., Chiumento, A., ... Ommeren, M. (2015). Problem Management Plus (PM+) for common mental disorders in a humanitarian setting in Pakistan; study protocol for a randomised controlled trial (RCT). BMC Psychiatry, 15(1), 232.
- Smith, S. L., Misago, C. N., Osrow, R. A., Franke, M. F., Iyamuremye, J. D., Dusabeyezu, J. D., ... Raviola, G. J. (2017). Evaluating process and clinical outcomes of a primary care mental health integration project in rural Rwanda: A prospective mixed-methods protocol. *BMJ Open*, 7(2), e014067.
- Spain, D., Sin, J., Paliokosta, E., Furuta, M., Prunty, J. E., Chalder, T., ... Happé, F. G. (2017). Family therapy for autismspectrumdisorders (Review). Cochrane Database of Systematic Reviews, 2017(5), 011894.

-WILEY-C Campbell Collaboration

26 of 34

- Susan, S., & Juillerat, S. (2012). The impact of a low cost wheelchair on the quality of life of the disabled in the developing world. *International Medical Journal of Experimental and Clinical Research*, 18(9), CR533.
- Thai, L., & Yasin, M. H. M. (2016). Magic finger teaching method in learning multiplication facts among deaf students. *Journal of Education* and Learning, 5(3), 40–50.
- Travis, J., & Geiger, M. (2010). The effectiveness of the Picture Exchange Communication System (PECS) for children with autism spectrum disorder (ASD): A South African pilot study [corrected] [published erratum appears in Child Lang Teach Ther 2010;26(3):383-384]. Child Language Teaching & Therapy, 26(1), 39–59.
- Tripney, J., Roulstone, A., Vigurs, C., Hogrebe, Nina, Schmidt, E., & Stewart, R. (2015). Interventions to improve the labour market situation of adults with physical and/or sensory disabilities in low- and middle-income countries: A systematic review. *Campbell Systematic Reviews*, 11(20).
- Ugwuanyi, L. T., & Adaka, T. A. (2015). Effect of auditory training on reading comprehension of children with hearing impairment in Enugu State. International Journal of Special Education, 30(1), 58–63.
- Valentini, N. C., & Rudisill, M. E. (2004). An inclusive mastery climate intervention and the motor skill development of children with and without disabilities. *Adapted Physical Activity Quarterly*, 21(4), 330–347.
- Vally, Z., & Abrahams, L. (2016). The effectiveness of peer-delivered services in the management of mental health conditions: A metaanalysis of studies from low- and middle-income countries. *International Journal for the Advancement of Counselling*, 38(4), 330–344.
- Van, G., Nadja, Tharyan, P., Lewin, S., Rao, G. N., Meera, S. M., ... Patel, V. (2013). Non-specialist health worker interventions for the care of mental, neurological and substance-abuse disorders in low-and middle-income countries. *The Cochrane Library*.
- Varghese, A. R., Rajagopalan, K., & Freejo, K. J. (2011). Impact of including videos of inspiring differentially abled personalities on developing attitude towards disabled students among high school students. *Learning Community-An International Journal of Educational and Social Development*, 2(2), 307–311.
- Vatandoost, N., Yarmohammadian, A., Abedi, A., Ghaziasgar, N., & Moghtadaie, M. (2013). The effect of auditory perception training on reading performance of the 8-9-year old female students with dyslexia: A preliminary study. *Audiology*, 22(4), 60–68.
- Velema, J. P., Ebenso, B., & Fuzikawa, P. L. (2008). Evidence for the effectiveness of rehabilitation-in-the-community programmes. *Leprosy Review*, 79, 65–82.
- Verhaeghe, N., De, M., Jan, Maes, L., Van, H., Cornelis, & Annemans, L. (2011). Effectiveness and cost-effectiveness of lifestyle interventions on physical activity and eating habits in persons with severe mental disorders: A systematic review. The International Journal of Behavioral Nutrition and Physical Activity, 8(1), 28.
- Vigneswara, PI, Kumar, AK, & Aswani, R. 2017. Employment outcomes of skills training in South Asian countries: An evidence summary. EPPI-Centre.
- Vigneswara, PI, Kumar, AK, & Aswani, R. 2017. Employment Outcomes of Skills Training in South Asian Countries: An Evidence Summary. Department for International Development.
- Virendrakumar, B., Jolley, E., Badu, E., & Schmidt, E. (2018). Disability inclusive elections in Africa: A systematic review of published and unpublished literature. *Disability & Society*, 33(4), 509–538.
- Wade, C., Llewellyn, G., & Matthews, J. (2008). Review of parent training interventions for parents with intellectual disability. *Journal of Applied Research in Intellectual Disabilities*, 21(4), 351–366.
- Wahl, H. W., Fänge, A., Oswald, F., Gitlin, L. N., & Iwarsson, S. (2009). The home environment and disability-related outcomes in aging individuals: What is the empirical evidence? *The Gerontologist*, 49(3), 355–367.

- Wallander, J. L., Bann, C. M., Biasini, F. J., Goudar, S. S., Pasha, O., Chomba, E., ... Carlo, W. A. (2014). Development of children at risk for adverse outcomes participating in early intervention in developing countries: A randomized controlled trial. *Journal of Child Psychology and Psychiatry*, 55(11), 1251–1259.
- Wang, P. (2008). Effects of a parent training program on the interactive skills of parents of children with autism in China. *Journal of Policy and Practice in Intellectual Disabilities*, 5(2), 96–104.
- Wapling, L. 2016. A systematic literature review of education systems in low-and middle income countries commissioned by CBM. CBM.
- Weiden, P. J., Manning, R., Wolfgang, C. D., Ryan, J., Mancione, L., Han, G., ... Mayo, M. G. (2016). A randomized trial of iloperidone for prevention of relapse in schizophrenia: The REPRIEVE study. CNS Drugs, 30(8), 735–747.
- Weinmann, S., Roll, S., Schwarzbach, C., Vauth, C., & Willich, S. N. (2010). Effectiveness of Ginkgo biloba (ginkgo) in dementia a systematic review. BMC Geriatrics, 10(1), 25–26.
- Westbrook, J. D., Fong, C. J., Nye, C., Williams, A., Wendt, O., & Cortopassi, T. (2013). Pre-graduation transition services for persons with autism spectrum disorders: Effects on employment outcomes. *Campbell Systematic Reviews*, 9(11), 1–70.
- White, H., Saran, A., Polack, S., & Kuper, H. (2018). A uniaxial load frame for in situ neutron studies of stress-induced changes in cementitious materials and related systems. Improving social inclusion and empowerment for people with disabilities in low- and middle-income countries: why does it matter and what works? 89, 092903.
- Wolmer, L., Laor, N., Dedeoglu, C., Siev, J., & Yazgan, Y. (2005). Teacher-Mediated Intervention after Disaster: A Controlled Three-Year Follow-Up of Children's Functioning. *Journal of Child Psychology and Psychiatry*, 46(11), 1161–1168.
- World Health Organisation (WHO) (2010). Community Based Rehabilitation Matrix.
- World Health Organisation (2014). Eliminating forced, coercive and otherwise involuntary sterilization: An interagency statement, OHCHR, UN Women, UNAIDS, UNDP, UNFPA, UNICEF and WHO.
- World Heath Organisation, & International Spinal Cord Society (2013). International perspectives on spinal cord injury. World Health Organisation.
- Wu, J., Wang, Y., & Wang, Z. (2017). The effectiveness of massage and touch on behavioural and psychological symptoms of dementia: A quantitative systematic review and meta-analysis. *Journal of Advanced Nursing*, 73, 2283–2295.
- Xie, C. L., Wang, X. D., Chen, J., Lin, H. Z., Chen, Y. H., Pan, J. L., & Wang, W.
 W. (2015). A systematic review and meta-analysis of cognitive behavioral and psychodynamic therapy for depression in Parkinson's disease patients. *Neurological ScienceS*, *36*(6), 833–843.
- Yawovi, T. 2009. Impact of rehabilitation care on the social inclusion of people with disabilities in Togo. Handicap International.
- Yilmaz, M., Sari, H. Y., Serin, G. E. Ç., Kisa, S. S., & Aydin, Ö. (2014). The effectiveness of nutrition and activity programmes for young adults with intellectual disabilities. *International Journal of Caring Sciences*, 7(2), 449–459.
- Yoon, I. A., Slade, K., & Fazel, S. (2017). Outcomes of psychological therapies for prisoners with mental health problems: A systematic review and meta-analysis. *Journal of Consulting and Clinical Psychology*, 85(8), 783–802.
- Yu, H., Liu, Y., Li, S., & Ma, X. (2009). Effects of music on anxiety and pain in children with cerebral palsy receiving acupuncture: A randomized controlled trial. *International Journal of Nursing Studies*, 46(11), 1423–1430.
- Zanon, M. A., Porfírio, G. J. M., Riera, R., & Martimbianco, A. L. C. (2015). Neurodevelopmental treatment approaches for children with cerebral palsy. Cochrane Database of Systematic Reviews (11).
- Zhang, X., Reinhardt, J. D., Gosney, J. E., & Li, J. (2013). The NHV rehabilitation services program improves long-term physical

functioning in survivors of the 2008 Sichuan earthquake: a longitudinal quasi experiment. PLoS One, 8(1), e53995.

Zuurmond, M., O'Banion, D., Gladstone, M., Carsamar, S., Kerec, M., Baltussen, M., ... Polack, S. (2018). Evaluating the impact of a community-based parent training programme for children with cerebral palsy in Ghana. PLoS One, 13(9), e0202096.

References to ongoing studies

- Bangpan, M., Dickson, K., Felix, L., & Chiumento, A. (2017). The impact of mental health and psychosocial support interventions on people affected by humanitarian emergencies: A systematic review humanitarian evidence programme. Humanitarian Evidence Programme. Oxford: Oxfam GB.
- Lietz, P., Kos, J., Dix, K., Trevitt, J., Uljarevic, M., & O'Grady, E. (2018). Interventions for anxiety in school-aged children with autism spectrum disorder (ASD).
- Fleming, P, Furlong, M, McGilloway, S, Keogh, F, Hernon, M, & Stainton, T. (2016). Personal budgeting interventions to improve health and social care outcomes for people with a disability. Campbell systematic reviews
- Zanon, M. A., Porfírio, G. J. M., Riera, R., & Martimbianco, A. L. C. (2015). Neurodevelopmental treatment approaches for children with cerebral palsy. Cochrane Database of Systematic Reviews (11).

References to Excluded studies

- Abolfazli, R., Hosseini, M., Ghanizadeh, A., Ghaleiha, A., Tabrizi, M., Raznahan, M., ... Akhondzadeh, S. (2011). Double-blind randomized parallel-group clinical trial of efficacy of the combination fluoxetine plus modafinil versus fluoxetine plus placebo in the treatment of major depression. Depression and Anxiety, 28, 297-302.
- Acarturk, C., Konuk, E., Cetinkaya, M., Senay, I., Sijbrandij, M., Gulen, B., & Cuijpers, P. (2016). The efficacy of eye movement desensitization and reprocessing for post-traumatic stress disorder and depression among Syrian refugees: results of a randomized controlled trial. Psychological Medicine, 46, 2583-2593.
- Akhondzadeh, S., Milajerdi, M. R., Amini, H., & Tehrani-Doost, M. (2006). Allopurinol as an adjunct to lithium and haloperidol for treatment of patients with acute mania: A double-blind, randomized, placebocontrolled trial. Bipolar Disorders, 8, 485-489.
- Aliyev, Z. N., & Aliyev, N. A. (2008). Valproate treatment of acute alcohol hallucinosis: A double-blind, placebo-controlled study. Alcohol and Alcoholism, 43, 456-459.
- Alsaraireh, F. A., & Aloush, S. M. (2017). Mindfulness meditation versus physical exercise in the management of depression among nursing students. Journal of Nursing Education, 56, 599-604.
- Amidfar, M., Khiabany, M., Kohi, A., Salardini, E., Arbabi, M., Roohi Azizi, M., ... Akhondzadeh, S. (2017). Effect of memantine combination therapy on symptoms in patients with moderate-to-severe depressive disorder: Randomized, double-blind, placebo-controlled study. Journal of Clinical Pharmacy & Therapeutics, 42, 44-50.
- Baker-Henningham, H., Powell, C., Walker, S., & Grantham-McGregor, S. (2005). The effect of early stimulation on maternal depression: a cluster randomised controlled trial. Archives of Disease in Childhood, 90(12), 1230-1234.
- Brooke-Sumner, C., Petersen, I., Asher, L., Mall, S., Egbe, C. O., & Lund, C. (2015). Systematic review of feasibility and acceptability of psychosocial interventions for schizophrenia in low and middle income countries. BMC Psychiatry, 15(1), 19.
- Chan, A. T., Sun, G. Y., Tam, W. W., Tsoi, K. K., & Wong, S. Y. (2017). The effectiveness of group-based behavioral activation in the treatment of depression: An updated meta-analysis of randomized controlled trial. Journal of Affective Disorders, 208, 345-354.

- Collaboration -WILEY Fried. L. P., & Guralnik, J. M. (1997). Disability in older adults: Evidence regarding significance, etiology, and risk. Journal of the American Geriatrics Society, 45(1), 92-100.
- Van Niekerk, L., Coetzee, Z., Engelbrecht, M., Landman, S., Motimele, M., Terreblanche, S., & Hajwani, Z. (2011). Supported employment: Recommendations for successful implementation in South Africa. South African Journal of Occupational Therapy, 41, 85-90.
- Young, L., Sheets, D., & Paul, K. (2013). Effectiveness of physical activity interventions with healthy older women: A systematic review protocol. JBI Database of Systematic Reviews & Implementation Reports. 11. 77-87.
- Zheng, W., Xiang, Y.-Q., Ungvari, G. S., Chiu, H. F., Ng, C. H., Wang, Y., & Xiang, Y.-T. (2016). Huperzine A for treatment of cognitive impairment in major depressive disorder: A systematic review of randomized controlled trials. Shanghai Archives of Psychiatry, 28, 64-71.

ADDITIONAL REFERENCES

- ADB. (2005). Disability Brief: Identifying and addressing the needs of disabled people. Asian Development Bank. Retrieved from http://www2.adb. org/Documents/Reports/Disabled-People-Development/disabilitybrief.pdf
- Andresen, E. M., Lollar, D. J., & Meyers, A. R. (2000). Disability outcomes research: Why this supplement, on this topic, at this time? Archives of Physical Medicine and Rehabilitation, 81, S1-S4.
- Banks, & Polack. (2015). The Economic Costs of Exclusion and Gains of Inclusion of People with disabilities: Evidence from Low and Middle Income Countries
- Banks, L. M., Kuper, H., & Polack, S. (2017). Poverty and disability in lowand middle-income countries: A systematic review. PLoS One, 12(12).
- CBM. (2012). A human rights-based approach to disability in development. Christian Blind Mission, CBM. Retrieved from http://www.cbm.org/Ahuman-rights-based-approach-to-disability-in-development-396302. php
- CRC. (1989). Convention on the Rights of the Child. Retrieved from http:// www.ohchr.org/EN/ProfessionalInterest/Pages/CRC.aspx
- Deke, J., Emily Sama-Miller, E., & Alan Hershey, A. (2015). Addressing Attrition Bias in Randomized Controlled Trials: Considerations for Svstematic Evidence Reviews. Research and Evaluation. Administration for Children and Families, U.S. Department of Health and Human Service. Retrieved from https://homvee.acf.hhs.gov/ HomVEE-Attrition-White_Paper-7-2015.pdf
- DFID. (2000). Disability, poverty and development. UK: Department for International Development.
- Fleming, P., McGilloway, S., Hernon, M., Furlong, M., O'Doherty, S., Keogh, F., & Stainton, S. (2019). Individualised funding interventions to improve health and social care outcomes for people with a disability: a mixedmethods systematic review. Campbell Systematic Reviews. Retrieved from https://campbellcollaboration.org/media/k2/attachments/0268_ DCG_Fleming_Individualised_funding.pdf
- Glickman, S. W., Anstrom, K. J., Lin, L., Chandra, A., Laskowitz, D. T., Woods, C. W., & Freeman, D. H. (2008). Challenges in enrollment of minority, pediatric, and geriatric patients in emergency and acute care clinical research. Annals of Emergency Medicine, 51(6), 775-780.
- Groce, N., Kett, M., Lang, R., & Trani, J. F. (2011). Disability and poverty: The need for a more nuanced understanding of implications for development policy and practice. Third World Quarterly.
- Hammerstrøm, K., Wade, A., Jørgensen, A. K., & Hammerstrøm, K. (2010). Searching for studies. Education, 54(11-3).
- Hróbjartsson, A., Forfang, E., Haahr, M., Als-Nielsen, B., & Brorson, S. (2007). Blinded trials taken to the test: an analysis of randomized clinical trials that report tests for the success of blinding. International Journal of Epidemiology, 36(3), 654-663.

VILEY-C Campbell Collaboration

- Iemmi, V., Gibson, L., Blanchet, K., Kumar, S., Rath, S., Hartley, S., & Kuper, H. (2015). Community-based rehabilitation for people with disabilities in low-and middle-income countries: a systematic review. *Campbell Systematic Reviews*, 15.
- Kuper, H., Saran, A., & White, H. (2018). Rapid Evidence Assessment (REA) of What Works to Improve Educational Outcomesfor People with Disabilities in Low- and Middle-Income Countries. Department of International Development.
- Mendis, P., Gunnel, N., Ann, G., & Einar, H. (1989). Training in the community for people with disabilities. World Health Organisation.
- Metts, R., & Mondiale, B. (2004). Disability and Development, background paper for the World Bank. World Bank. Retrieved 2017, from http:// siteresources.worldbank.org/DISABILITY/Resources/280658-1172606907476/mettsBGpaper.pdf
- Mitra, S. (2013). A data revolution for disability-inclusive development. Lancet Global Health.
- Ramey, D., Lydon, S., Healy, O., McCoy, A., Holloway, J., & Mulhern, T. (2016). A systematic review of the effectiveness of precision teaching for individuals with developmental disabilities. *Review Journal of Autism and Developmental Disorders*, 3(3), 179–195.
- Trani, J. F., & Loeb, M. (2012). Poverty and disability: A vicious circle? Evidence from Afghanistan and Zambia. Journal of International Development.
- Tripney, J., Roulstone, A., Vigurs, C., Hogrebe, N., Schmidt, E., & Stewart, R. (2015). Interventions to improve the labour market situation of adults with physical and/or sensory disabilities in low-and middle-income countries. Campbell Systematic Reviews.
- UNCRPD. (2006). The United Nation Convention on the Rights of Person with Disabilities. Retrieved from https://www.un.org/development/desa/ disabilities/convention-on-the-rights-of-persons-with-disabilities/ convention-on-the-rights-of-persons-with-disabilities-2.html
- UNDP. (2010). Inclusive development. United Nations Development Program.
- United Nation., 1. (1994). Standard Rules on the Equalization of Opportunities for Persons with Disabilities. Retrieved from https:// www.un.org/disabilities/documents/gadocs/standardrules.pdf
- United Nations General Assembly., 2. (2006). Convention on the Rights of Persons with Disabilities A/RES/61/106 Session 61 resolution 106.
- USAID. (1997). USAID Disability. U.S. Agency for International Development.
- Ustun, T., Chatterji, S., Mechbal, A., & Murray, C. L. (2003). WHS Collaborating groups: The World health surveys. In C. J. L. Murray &

D. Evans (Eds.), *Health systems performance assessment: debates, methods and empiricism.* Geneva: World Health Organisation.

- Velema, J. P., Ebenso, B., & Fuzikawa, P. L. (2008). Evidence for the effectiveness of rehabilitation-in-the-community. *Leprosy Review*, 79(1), 65.
- Walsh, F. P., & Tickle, A. C. (2013). Working towards recovery: the role of employment in recovery from serious mental health problems: a qualitative meta-synthesis. *International Journal of Psychosocial Rehabilitation*, 17(2), 49–97.
- White, H. (2019). The twenty-first century experimenting society: the four waves of evidence revolution. *Palgrave Communications*, 5(1), 47.
- White, H., Saran, A., & Kuper, H. (2018). Improving social inclusion and empowerment for people with disabilities in low- and middle-income countries: why does it matter and what works? Department of International Development.
- WHO, I. (2004). UNESCO: CBR: A strategy for rehabilitation, equalization of opportunities, poverty reduction and social inclusion of people with disabilities. Geneva: WHO.
- World Bank., 2. (2008). Project appraisal document on a proposed credit to the People's Republic of Bangladesh for. Washington: World Bank. Retrieved from http://documents.worldbank.org/curated/en/ 485601468003560874/pdf/

417950PAD0P1061ly100IDAR20081021911.pdf

- World Health Organisation. (2011). World Report on Disability. Geneva: World Health Organisation.
- World Bank. (2017). World Bank country and lending groups. Retrieved from https://datahelpdesk.worldbank.org/knowledgebase/articles/906519world-bank-country-and-lending-groups
- WPA. (1982). World Programme of Action Concerning Disabled Persons. Retrieved from https://www.un.org/development/desa/disabilities/ resources/world-programme-of-action-concerning-disabled-persons. html

How to cite this article: Saran A, White H, Kuper H. Evidence and gap map of studies assessing the effectiveness of interventions for people with disabilities in low-and middleincome countries. *Campbell Systematic Reviews*. 2020;16:e1070. https://doi.org/10.1002/cl2.1070

APPENDIX A



| IABLE A1 Description of methods used for inclusion and exclusion of studi | ription of methods used for inclusion and exclusion of st | tudies |
|--|---|--------|
|--|---|--------|

| Selection criteria | Inclusion | Exclusion |
|-----------------------|---|---|
| Publication year | After 2000 | Before 2000 |
| Publication status | Completed and on-going | None |
| Study design | The EGM will include systematic reviews of effects of interventions and effectiveness studies that used either: (a) randomised experimental design, or (b) rigorous quasiexperimental design, (c) natural experiments, (d) regression discontinuity, (e) propensity score matching, (f) difference in difference, (g) instrumental variables, (h) other matching design and (i) single subject design | Literature reviews, noneffectiveness studies, case studies and qualitative studies |
| Population | People with disability, and/or their family, their caregivers, their community living in low- and middle-income countries Disability is defined as impairments, activity limitations, and participation restrictions denoting the negative aspects of the interaction between an individual (with a health condition) and that individual's contextual factors (environmental and personal factors) (World Health Organisation, 2011) For Impact evaluation we will include participants from low- and middle-income countries only, as this was the original commitment of CBR (Helander, 1989) | People with disabilities and/or their family, their caregivers, their community living in high-income countries |
| Interventions | We will include effectiveness studies. A CBR programme is formed by one or more activities in one or more of the five components (health, education, livelihood, social and empowerment). List of activities for each element of the five components are presented within the CBR guidelines under the section "Suggested activities". The following activities are here given as examples: • Health: training PWD in the use of assistive devices; providing information to PWD and their family or their caregivers about time and location of activities for screening health conditions and impairments associated with disabilities • Education: providing education and training for families or caregivers of PWD; installing ramps in schools to make them accessible to PWD using wheelchairs • Livelihood: linking the jobseeker with disability to existing support services; advocating before relevant public and private agencies to ensure accessible housing for PWD • Social: converting information to PWD about the sports opportunities available within the community • Empowerment: helping PWD running meetings of new self-help group; involving disabled's people organizations in CBR planning, implementation, and monitoring | All the clinical trials, interventions for reversible form of illness will be excluded. Interventions not focused on people with disabilities. We will also exclude studies that deals temporary or reversible form of disability for examples, maternal depression or back pain |
| Outcome | We will use the CBR framework for outcomes. | None |
| Quality | We will not restrict based on quality | None |

ILEY-C Campbell Collaboration

APPENDIX B

TABLE B1 List of databases

Indexes

International organizations

- ILO
- DFID (including Research for Development [R4D])
- UNESCO
- WHO
- Disability Programme of the United Nations Economic and Social Commission for Asia and the Pacific (UNSCAP)
- United States Agency for International Development (USAID)
- Evidence and gap map databases
- International Initiative for Impact Evaluation (3ie) Evidence and gap map repository
- Swedish Agency For Health Technology Assessment and Assessment of Social Services
- Collaboration for Environmental Evidence
- Global Evidence Mapping Initiative
- Evidence based Synthesis Program (Department of Veteran Affairs)
- Cochrane
- Evidence based policing matrix
- EPPI Centre Evaluation Database of Education Research

Systematic review databases

- Cochrane
- Campbell
- 3ie Systematic Review Database
- Research for Development
- Epistemonikos

Academic databases

- Econlit
- The National Bureau of Economic Research (NBER)
- Social Science Research Network (SSRN)
- International Bibliography of Social Sciences (IBSS)
- Applied Social Sciences Index and Abstracts (ASSIA)
- Embase
- PsycINFO
- MEDLINE
- WHO's Global Health Library
- CABI's Global Health
- ERIC
- CINHAL
- SCOPUS
- Web of Science

- Other websites
- Humanity and Inclusion (HI) http://www.hi-us.org/publications
- CBM https://www.cbm.org/Publications-252011.php
- Plan international https://plan-international.org/publications
- PAHO https://www.paho.org/hq/index.php?lang=en
- UNICEF www.unicef.org/
- UNICEF Innocenti Research Centre www.unicef-irc.org/
- UN Women http://www.unwomen.org/en/digital-library/ publications
- UNESCO http://www.unesco.org/library/e-resDatabases1.html
- United Nations Population Fund http://digitallibrary.un.org/ record/703986
- UN Economic and Social Council www.un.org/en/ecosoc/
- IRC https://www.rescue.org/reports-and-resources
- IFRC https://www.ifrc.org/en/publications-and-reports/
- CARE www.care.org
- Indian Citation Index (ICI) http://www.indiancitationindex.com/
- Save the Children www.savethechildren.org
- British Library for Development Studies http://blds.ids.ac.uk/
- ELDIS http://www.eldis.org/
- Essential Health Links http://www.healthnet.org/essential-healthlinks
- Global Health and Global Health Archive http://www.cabi.org/ datapage.asp?iDocID=169
- African developmental bank https://www.afdb.org/en/knowledge/ african-development-institute/information-and-library-services/
- Young Lives https://www.younglife.org/ForEveryKid/YoungLives/ Pages/default.aspx
- Association for the Development of Africa http://www.adeanet. org/en/knowledge-and-resources
- Médians Sans Frontières www.msf.org.uk/ and http:// fieldresearch.msf.org/msf/
- Action Aid http://www.actionaid.org/publications
- ILO https://www.ilo.org/global/publications/lang--en/index.htm
 DFID (including Research for Development (R4D) https://www.gov.uk/dfid-research-outputs
- Disability Programme of the United Nations Economic and Social Commission for Asia and the Pacific (UNSCAP) https://www. unescap.org/our-work/social-development/disability
- United States Agency for International Development (USAID) https://www.usaid.gov/kyrgyz-republic/key-documents
- World Vision www.worldvision.org.uk/
- Department for International Development www.dfid.gov.uk/
- World Food Programme https://www.wfp.org/evaluation/list (Evaluations)
- Valid International www.validinternational.org/
- Concern Worldwide www.concern.net/
- International Red Cross/Red Crescent www.ifrc.org/

(Continues)

APPENDIX C

TABLE C1 Coding tool



| Category | | Answer |
|--------------------------|--|---|
| Descriptive information | Title Author citation Publication date URL Volume no Issue no | Open answer Open answer Open answer Open answer Open answer Open answer |
| Geographical information | World Bank region Country | South Asia Sub-Saharan Africa East Asia and Pacific Europe and Central Asia Latin America and Caribbean Middle East and North Africa North America Low income Lower Middle income Upper Middle income Conflict affected |
| Study design | | See relevant country list as per World Bank Region - Systematic reviews - RCT - Quasiexperimental study - Case-control - Cohort - Controlled trial |
| Population | | Children Adults Elderly Women Men LGBT community Conflict affected Disadvantaged Migrants/refugees Ethnic minorities |
| Type of Impairment | | Physical impairment Visual impairment Mental impairment Hearing impairment Intellectual/learning impairment |
| Intervention | | Health Promotion Prevention Medical care Rehabilitation Assistive devices Education Early child development Nonformal Primary and secondary Lifelong learning Livelihood Skills development Self-employment Waged employment Financial services Social protection |

32 of 34 WILEY-Campbell _____Collaboration

SARAN ET AL.

| Category | Answer |
|----------|---|
| | Social Relationship, marriage and family Personal assistance Culture, religion and arts Sports, recreation and leisure Access to justice Empowerment Social mobilisation Political Participation Language and communication Self-help groups and Disabled People Organisation Advocacy and Governance Advocacy and Governance |
| Outcome | Health Mental health and cognitive development Access to health services Immunization Health check-up Rehabilitation Access to assistive devices Nutrition Morbidity and mortality Education Enrolment to primary, secondary and tertiar education Attendance Education in mainstream education facilities inclusive education Social and life skill development Access to educational services Livelihood Employment in formal and informal sector Access to job market Control over own money Access to social protection programs Participation in development of inclusive policies Social Stigma and discrimination Safety Participation in mainstream recreational, leisur and sports activity legal rights Access to justice Participation in cultural and religious activity |

- Empowerment
 - Informed choices
 - O Positions in public institutions and Judiciary
 - Voting rights
 - Representation at community level
- Advocacy

APPENDIX D



TABLE D1 Search strategy

Search string/key words (For ovid medline platform)

Developing Country Free Text

- (developing OR less-developed OR less* developed OR "under developed" OR underdeveloped OR under-developed OR middle-income OR
 "middle income" OR "low income" OR low-income OR underserved OR "under served" OR deprived or poor*) adj3 (countr* OR nation OR
 population OR world OR state OR economy OR economies).mp
- ("third world" OR L&MIC OR L&MIC OR LAMIC OR LDC OR LIC OR LMIC* OR lami countr* OR transitional countr*).mp
- (Africa OR "Sub-Saharan Africa" OR "North Africa" OR "West Africa" OR "East Africa" OR Algeria OR Angola OR Benin OR Botswana OR Burkina Faso OR Burundi OR Cameroon OR "Cape Verde" OR "Central African Republic" OR Chad OR "Democratic Republic of the Congo" OR "Republic of the Congo" OR Congo OR "Cote d'Ivoire" OR "Ivory Coast" OR Djibouti OR Egypt OR "Equatorial Guinea" OR Eritrea OR Ethiopia OR Gabon OR Gambia OR Ghana OR Guinea OR Guinea-Bissau OR Kenya OR Lesotho OR Liberia OR Libya OR Madagascar OR Malawi OR Mali OR Mauritania OR Morocco OR Mozambique OR Namibia OR Niger OR Nigeria OR Rwanda OR "Sao Tome" OR Principe OR Senegal OR "Sierra Leone" OR Somalia OR Somaliand OR "South Africa" OR "South Sudan" OR Sudan OR Swaziland OR Tanzania OR Togo OR Tunisia OR Uganda OR Zambia OR Zimbabwe).mp.
- ("South America" OR "Latin America" OR "Central America" OR Mexico OR Argentina OR Bolivia OR Brazil OR Chile OR Colombia OR Ecuador OR Guyana OR Paraguay OR Peru OR Suriname OR Uruguay OR Venezuela OR Belize OR "Costa Rica" OR "El Salvador" OR Guatemala OR Honduras OR Nicaragua OR Panama).mp.
- ("Middle East" OR "South-East Asia" OR "Indian Ocean Island*" OR "South Asia" OR "Central Asia" OR Caucasus OR Afghanistan OR Azerbaijan OR Bangladesh OR Bhutan OR Burma OR Cambodia OR China OR Georgia OR India OR Iran OR Iraq OR Jordan OR Kazakhstan OR Korea OR "Kyrgyz Republic" OR Kyrgyzstan OR Lao OR Laos OR Lebanon OR Macao OR Mongolia OR Myanmar OR Nepal OR Oman OR Pakistan OR Russia OR "Russian Federation" OR "Saudi Arabia" OR Bahrain OR Indonesia OR Malaysia OR Philippines OR Sri Lanka OR Syria OR "Syrian Arab Republic" OR Tajikistan OR Thailand OR Timor-Leste OR Timor OR Turkey OR Turkmenistan OR Uzbekistan OR Vietnam OR "West Bank" OR Gaza OR Yemen OR Comoros OR Maldives OR Mauritius OR Seychelles).mp.
- ("Pacific Islands" OR "American Samoa" OR Fiji OR Guam OR Kiribati OR "Marshall Islands" OR Micronesia OR New Caledonia OR "Northern Mariana Islands" OR Palau OR "Papua New Guinea" OR Samoa OR "Solomon Islands" OR Tonga OR Tuvalu OR Vanuatu).mp

Systematic review key words

 - ((systematic* or synthes*) adj3 (research or evaluation* or finding* or thematic* or report or descriptive or explanatory or narrative or meta* or review* or data or literature or studies or evidence or map or quantitative or study or studies or paper or impact or impacts or effect* or compar*)).ti,ab,sh.

OR

("meta regression" or "meta synth*" or "meta-synth*" or "meta analy*" or "metaanaly*" or "meta-analy*" or "metanaly*" or "metaregression" or "metanaly*" or "metaregression" or "methodologic* overview" or "pool* analys*" or "pool* data" or "quantitative* overview" or "research integration").ti,ab,sh.

OR

(review adj3 (effectiveness or effects or systemat* or synth* or integrat* or map* or methodologic* or quantitative or evidence or literature)).ti,ab,sh.

Qualitative review search term

((("meta ethnography" OR "meta ethnographic") OR ("meta synthesis") OR (synthesis AND ("qualitative literature" OR "qualitative research")) OR ("critical interpretive synthesis") OR ("systematic review" AND ("qualitative research" OR "qualitative literature" OR "qualitative studies")) OR ("thematic synthesis") OR ("realist review" OR "realist synthesis") OR ((("qualitative systematic review" OR "qualitative evidence synthesis")) OR ("qualitative systematic review" OR "qualitative evidence synthesis")) OR (("qualitative systematic reviews" OR "qualitative evidence synthesis")) OR (("qualitative systematic reviews" OR "qualitative evidence syntheses")) OR (("qualitative research" OR "qualitative literature" OR "qualitative literature" OR "qualitative systematic reviews")) OR (("qualitative research" OR "qualitative literature" OR "qualitative studies")) OR (("qualitative research" OR "qualitative literature" OR "qualitative studies")) OR (("literature search" OR "literature searching" OR "literature searchs")) OR ("qualitative research" OR "qualitative literature" OR "qualitative studies")) OR ((Noblit AND Hare)) OR ("meta narrative" OR "meta narratives" OR "meta narratives")) OR ("meta narratives")) OR ("meta narrative" OR "meta narratives")) OR ("meta narrat

Disability key words

- ((Disable* or Disabilit* or Handicapped) adj5 (person* or people or child*or adolescen* or women or mother*or maternal, group)).sh,ti,ab.
- ((physical* or intellectual* or learning or psychiatric* or sensory or motor or neuromotor or cognitive or mental* or developmental or communication or learning) adj2 (disabilit* or disabl* or handicap*)).ti,ab
- ((cognitive* or learning or mobility or sensory or visual* or vision or sight or hearing or physical* or mental* or intellectual*) adj2 impair*).ti,ab
- ((mental health or mental disorder* or depress* or anxiety or psychiat* or well-being or quality of life or self-esteem or self perception)). ti,ab
- ((mental* or emotional* or psychiatric or neurological or neurologic) adj2 (disorder* or ill or illness*)).ti,ab (deaf or deafness or blind or blindness).ti,ab
- exp Disabled persons/
- (Autis* or Dyslexi* or Down* Syndrome or Mongolism or Trisomy 21).sh,ti,ab.
- exp Intellectual disability/or exp Developmental Disabilities/or exp Child Development Disorders, Pervasive/or exp Communication Disorders/

TABLE D1 (Continued)

WILEY

Search string/key words (For ovid medline platform)

- ((Intellectual* or Educational* or Mental* or Psychological* or Developmental) adj5 (impair* or retard* or deficienc* or Deficien* or disable* or disabili* or handicap* or ill*)).sh,ti,ab.
- ((Hearing or Acoustic or Ear*) adj5 (loss* or impair* or deficienc* or disable* or disabili* or handicap*)).sh,ti,ab.
- ((Visual* or Vision or Eye*) adj5 (loss* or impair* or deficienc* or disable* or disabili* or handicap*)).sh,ti,ab.
- (Deaf* or Blind*).sh,ti,ab
- exp Cerebral palsy/or exp Spina Bifida Cystica/or exp Spina Bifida Occulta/or exp Muscular dystrophies/or exp Arthritis/or exp Osteogenesis
 Imperfecta/or exp Musculoskeletal Abnormalities/or exp Brain Injuries/or exp Amputation/or exp Clubfoot/or exp Poliomyelitis/or exp
 Paraplegia/or exp Hemiplegia/ or exp Stroke/
- (Cerebral pals* or Spina bifida or Muscular dystroph* or Arthriti* or Osteogenesis imperfecta or Musculoskeletal abnormalit* or Musculo-skeletal abnormalit* or Muscular abnormalit* or Skeletal abnormalit* or Limb abnormalit* or Brain injur* or Amputation* or Clubfoot or Poliomyeliti* or Paraplegi* or Paralys* or Paralyz* or Hemiplegi* or Stroke* or Cerebrovascular accident*).sh,ti,ab.
- (Physical* adj5 (impair* or deficienc* or disable* or disabili* or handicap*)).sh,ti,ab.

ampbell

Collaboration

Equity terms:

- ((social* or socio-economic or socioeconomic or economic or structural or material) adj3 (advantage* or disadvantage* or exclude* or exclusion or include* or inclusion or status or position or gradient* or hierarch* or class* or determinant*)).sh,ti,ab.
- (equit* or inequit* or inequalit* or disparit* or equality).sf,ti,ab.